Investigation of a High-Frequency Link Single-Stage Asymmetrical Multilevel Converter for Grid Integration of Renewable Energy Systems

A DISSERTATION SUBMITTED TO THE FACULTY OF THE GRADUATE SCHOOL OF THE UNIVERSITY OF MINNESOTA BY

Kartik V Iyer

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF Doctor of Philosophy

Ned Mohan

April, 2017

© Kartik V Iyer 2017 ALL RIGHTS RESERVED

Acknowledgements

First and foremost, I would like to thank my adviser, Prof. Ned Mohan for bestowing upon me this opportunity to pursue a PhD. I would also like to extend my heartfelt gratitude to Prof. William P. Robbins for all those technical discussions that I had with him during my masters' thesis. I would also like to thank Prof. Sairaj Dhople and Prof. Peter Seiler who agreed to be in my committee for the thesis defense.

I want to express my gratitude to Dr. Eric Severson and Dr. Kaushik Basu for their help and guidance during my initial years. I want to immensely thank them for all the technical ideas and fruitful discussions. I am thankful to my friends/roommates/labmates, Dr. Ashish Sahoo and Dr. Srikant Gandikota for being there all the time for everything. I want to thank Dr. Rohit Baranwal, Dr. Saurabh Tewari, Ruben Otero and Dr. Suvankar Biswas for all the technical discussions during the course of the project. I am really grateful to my labmates and Srivatsal Sharma for the effort they took in reviewing my technical papers. I also want to thank Varsha Padhee, Siddharth Raju and Santhosh Krishnamoorthi for the good times spent in Minneapolis. Also, lastly I thank all the other members in the PERL group and my friends.

I cannot miss out thanking my parents for their love and guidance. I want to especially thank my elder brother, Dr. Ganpathy Iyer and his wife, Dr. Vidya Viswanath without whose support and advice nothing would have materialized.

Thank you very much.

Dedication

To those who held me up over the years

Abstract

Different power electronic converter topologies have been investigated to integrate renewable energy systems to the grid. Cascaded multilevel converters with a high-frequency link have emerged as a viable candidate for such applications. Electrical isolation can be provided using a compact high-frequency transformer connected in the link thus avoiding a bulky line frequency transformer. The use of cascaded modules allows the generation of a multilevel voltage having low total harmonic distortion (THD) but increases the overall system size. In this thesis, a fifteen level high-frequency AC-link single-stage asymmetrical multilevel converter for grid integration is proposed. The single-stage conversion approach eliminates the DC-link capacitors, resulting in a reduced footprint. The asymmetrical module voltages are generated by the multi-winding transformer having unequal turns on each of the secondaries. This allows the generation of fifteen output voltage levels, using only three modules in each phase. A modified uni-polar modulation strategy is proposed to generate multilevel output voltage. A modified hybrid modulation technique using triangular level shifted carriers based on the high frequency link is also investigated for this topology. Unlike previously used technique, the proposed modulation shifts the dominant harmonics in the output voltage to the sidebands of multiples of twice the switching frequency, thereby reducing the output filter size.

However, the incorporation of the switch non-idealities requires the implementation of a commutation strategy for the single-stage conversion. A detailed circuit analysis showing different modes of operation to generate a specific output voltage level, taking into account the switch non-idealites is outlined in the thesis. The analysis aids in the real-time implementation of the converter and, it also explains the distortion in the ideal output voltage profile. The analysis in general can be used for any isolated single-stage converter. The effect of switch non-idealities on the output voltage is analyzed and two compensation techniques are developed to improve the voltage profile.

A multi-winding high-frequency transformer is a critical component in the proposed converter topology. A four-winding transformer is designed and characterized using Ansys 3-D Finite Element Analysis and Network Analyzer measurements. The presence of transformer leakage inductance will require a clamp circuit to dissipate the leakage energy during commutation. A detailed circuit analysis showing the various modes of operation considering both the switch and transformer non-idealities is presented.

Comprehensive analysis is done for ensuring the generation of balanced currents in case of a module failure. A look up table is provided to operate the proposed topology for all possible cases. The presented concepts are verified by simulation and further validated experimentally on a three-phase fifteen level converter prototype.

Contents

Α	cknov	wledge	ements	i
D	edica	tion		ii
A	bstra	ict		iii
$\mathbf{L}^{\mathbf{i}}$	ist of	Table	s	viii
$\mathbf{L}^{\mathbf{i}}$	ist of	Figur	es	ix
1	Intr	roduct	ion	1
	1.1	Conve	entional interface to integrate renewables to grid	1
	1.2	Conve	entional interface for all electrified ships	2
	1.3	Prope	sed interface using high-frequency transformers	3
	1.4	Modu	lation and commutation strategies	5
2	Hig	h-Freq	quency Link Single-Stage Converter	8
	2.1	Multi	level converter with high-frequency transformer (HFT) isolation	8
		2.1.1	Circuit configuration	8
		2.1.2	Evaluation of circuit configuration based on system complexity $\ . \ .$	9
		2.1.3	Description of the circuit configuration	12
3	Idea	al Moo	lulation	14
	3.1	Modu	lation-I: Modified Unipolar Modulation	14
		3.1.1	Modulation of primary side full-bridge converter $\ldots \ldots \ldots \ldots$	14
		3.1.2	Modulation of secondary side AC-AC converter modules	14
		3.1.3	Generation of specific output voltage level	16
	3.2	Modu	lation-II: Modified Hybrid Modulation	18
		3.2.1	Modulation of primary side converter $\ldots \ldots \ldots \ldots \ldots \ldots$	18
		3.2.2	Signal generation for switches of secondary side modules	18

		3.2.3	Generation of specific output voltage level	21				
	3.3	Harm	onic spectrum of voltage for both modulation techniques	23				
		3.3.1	Spectrum of module output voltage: Modulation-I	23				
		3.3.2	Spectrum of module output voltage: Modulation-II	24				
		3.3.3	Spectrum of output voltage	24				
4	Noi	n-Ideal	l Modulation	27				
	4.1	Modu	lation-I: Incorporation of switch non-idealities	27				
		4.1.1	Effect on output voltage profile	27				
		4.1.2	Glitch Compensation-I	31				
		4.1.3	Glitch Compensation-I (during reverse power flow)	34				
		4.1.4	Glitch Compensation-II	37				
	4.2	Modu	lation-II: Incorporation of switch non-idealities	39				
		4.2.1	Dead-time in the switches of primary side H-bridge	40				
		4.2.2	Four-step commutation for switches of AC-AC Module	40				
		4.2.3	Effect on output voltage profile-Switching Cycle Transition $\ . \ . \ .$	42				
5	Des	ign of	High-Frequency Four Winding Transformer	47				
	5.1	Trans	former design	47				
		5.1.1	Transformer specifications and core material	47				
		5.1.2	Winding design	48				
		5.1.3	Inductance computation	49				
6	\mathbf{Sys}	stem Level Analysis of Proposed Converter Topology 5						
	6.1	Closed	d loop control	52				
	6.2	Opera	ation under module failure	54				
	6.3	Filter	inductor design	59				
		6.3.1	Design procedure	59				
		6.3.2	Comparison of filter size for two modulation strategies	60				
7	Haı	rdware	e Setup and Experimental Results	61				
	7.1	Hardy	ware setup	61				
	7.2	Modu	lation-I: Simulation and experimental results	63				
		7.2.1	Results of Glitch Compensation	63				
		7.2.2	Results for a three-phase system	68				
		7.2.3	Results for bi-directional power flow	74				
		7.2.4	Results for operation under module failure	76				
	7.3	Modu	lation-II: Simulation and experimental results	78				

8	Hig	gh-Frequency Link Asymmetrical Converter 9						
	8.1	Conve	rter topology	90				
		8.1.1	Circuit description	90				
		8.1.2	Modulation strategy of input side converter $\ldots \ldots \ldots \ldots \ldots$	91				
		8.1.3	HFT voltage rating analysis based on DC link current of CSI $\ . \ . \ .$	92				
		8.1.4	Effect of transformer leakage inductance	93				
		8.1.5	Control of CSI	95				
	8.2	Simula	ation results	96				
9	Cor	nclusio	n and Future Work	98				
	9.1	Concl	usion	98				
9.2 Study of several isolated multilevel converter topologies								
	9.3	Future	e work	101				
R	efere	nces		102				

List of Tables

2.1	Switch count for 500 kW/7.2 kV interface $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	10
2.2	System Complexity	11
2.3	Normalized index value	11
3.1	Module and total phase-to-neutral output voltages $\ldots \ldots \ldots \ldots \ldots$	16
3.2	Modulation technique for different positive levels in the output voltage	19
4.1	Instances requiring Glitch Compensation-II	39
5.1	Transformer specifications	47
6.1	Solution with reduced balanced currents for some fault cases $\ldots \ldots \ldots$	56
7.1	Specifications	78
8.1	Modulation Strategy for Different Positive Levels for V_s	91
8.2	Transformer Power Rating variation with the Leakage Inductance	94
9.1	Case Study of Several Isolated Multilevel Converter Topologies	100

List of Figures

1.1	Conventional Topology	1
1.2	Electrified Ship Propulsion Systems.	2
1.3	Proposed Topology	3
1.4	Proposed configuration for ship electrification	5
2.1	Block diagram of proposed topology.	10
2.2	Output voltage profile and harmonic spectrum of various configurations	11
2.3	Comparison of various configurations	12
2.4	Detailed per-phase circuit diagram of the proposed topology	12
3.1	Modulation-I: Logic level diagram and ideal signals	16
3.2	Modulation-I: Fundamental cycle of output voltage waveform $\ldots \ldots \ldots$	18
3.3	Modulation-II: Logic level diagram and ideal signals	20
3.4	Modulation II: Signal generation for asymmetrical modules $\ldots \ldots \ldots$	21
3.5	Modulation-II: Fundamental cycle of output voltage waveform	22
3.6	Modulation-I: Harmonic spectrum of module output voltage	23
3.7	Modulation-II: Harmonic spectrum of module output voltage \ldots	25
3.8	Harmonic spectrum comparison of output voltages	26
4.1	Non-ideal switching signals of converter modules	28
4.2	Circuit diagram with non-ideal switches- Switching Cycle Transition \ldots .	29
4.3	Circuit diagram with Glitch Comensation-I- Switching Cycle Transition $\ . \ .$	31
4.4	Fundamental cycle of output voltage with non-ideal switches	33
4.5	Fundamental cycle of output voltage with Glitch Compensation-I	33
4.6	Fundamental cycle of output voltage (Reverse Power Flow) $\ldots \ldots \ldots$	34
4.7	Fundamental cycle of output voltage with Glitch Compensation-I and II $\ . \ .$	34
4.8	Circuit diagram during reverse power flow with Glitch Comensation-I $~$	35
4.9	Effect of compensation on output voltage and input current	36
4.10	Switching signals showing cause and effect of Glitch Compensation-II $% \left({{{\bf{G}}_{{\rm{B}}}}_{{\rm{B}}}} \right)$	37
4.11	Circuit diagram showing cause and effect of Glitch Compensation-II \ldots	38
4.12	Non-ideal switching signals of converter modules	41

4.13	Circuit diagram with non-ideal switches- Switching Cycle Transition $\ . \ . \ .$	44
5.1	High-frequency transformer	49
5.2	Prototype of the four winding transformer	49
5.3	Modeling of a four-winding transformer	51
5.4	Characterization of a four-winding transformer using network analyzer	51
6.1	Control Model	53
6.2	Closed-loop control in simulation	53
6.3	Dynamic behavior of converter topology during load change	54
6.4	Three phase voltages for unfaulted case	55
6.5	Three line-line voltages for unfaulted case	55
6.6	Three line currents for unfaulted case	55
6.7	Three phase voltages for case $\{\cancel{A}, \cancel{Z}, 1\}$ in phase- c	56
6.8	Three line-line voltages for case $\{\cancel{A}, \cancel{Z}, 1\}$ in phase- c	57
6.9	Three line currents for case $\{\cancel{A}, \cancel{Z}, 1\}$ in phase- c	57
6.10	Three phase voltages for case $\{4, 2, 1\}$ in phase- c	58
6.11	Three line-line voltages for case $\{\cancel{A}, 2, 1\}$ in phase- c	58
6.12	Three line currents for case $\{4, 2, 1\}$ in phase- c	59
7.1	Hardware prototype of proposed converter topology	61
7.2	Experimental Setup	62
7.3	Digital signals to control the hardware setup: Modulation I $\ . \ . \ . \ .$.	63
7.4	Output voltage profile with switch non-idealities using the hardware setup .	64
7.5	Digital signals incorporating Glitch Compensation-I: Modulation-I \ldots .	64
7.6	Experimental output voltage profile with Glitch Compensation-I $\ldots \ldots$	65
7.7	Digital signals for reverse power flow using the hardware setup $\ldots \ldots$	65
7.8	Output voltages and current profile during reverse power flow	66
7.9	Digital signals without Glitch Compensation-II: Modulation-I	66
7.10	Digital signals with Glitch Compensation-II: Modulation-I	67
7.11	Simulated transformer link voltages: Modulation-I $\ .\ .\ .\ .\ .\ .$.	68
7.12	Experimental transformer link voltages: Modulation-I $\ \ldots\ \ldots\ \ldots\ \ldots$	69
7.13	Simulated module output voltages: Modulation-I \hdots	69
7.14	Experimental module output voltages: Modulation-I $\ .\ .\ .\ .\ .$.	70
7.15	Experimental output voltages and currents: Modulation-I $\ \ldots\ \ldots\ \ldots$	70
7.16	Experimental line currents: Modulation-I	71
7.17	Simulated transformer link currents: Modulation-I	71
7.18	Experimental transformer link currents: Modulation-I $\ \ldots\ \ldots\ \ldots\ \ldots$	72
7.19	Simulated transformer link currents-Switching cycle: Modulation-I $\ .\ .\ .$	72
7.20	Experimental transformer link currents- Switching cycle: Modulation-I	73

7.21	Experimental results of power flow from DC to three-phase AC \ldots .	74
7.22	Experimental results of power flow from three-phase AC to DC \ldots .	75
7.23	Harmonic spectrum of output phase voltage, v_a : Modulation-I	75
7.24	Harmonic spectrum of output current: Modulation I $\ldots \ldots \ldots \ldots \ldots$	76
7.25	Experimental results for module failure: Modulation-I	77
7.26	Harmonic spectrum of the output voltage, v_c of the faulted phase- c	77
7.27	Harmonic spectrum of the output current, i_c of the faulted phase- c	78
7.28	Simulation results of freewheeling interval: Modulation-II	79
7.29	Experimental results of freewheeling interval: Modulation-II	80
7.30	Simulation results showing PWM operation: Modulation-II	80
7.31	Experimental results showing PWM operation: Modulation-II	81
7.32	Digital signals with Glitch Compensation-I: Modulation-II	81
7.33	Experimental output voltage profile with Glitch Compensation-I \ldots	82
7.34	Simulated output voltages and currents: Modulation II	83
7.35	Experimental output voltages and currents: Modulation II	84
7.36	Three phase line-to-neutral voltages: Modulation II	84
7.37	Three phase line-to-line voltages: Modulation II	85
7.38	Fundamental cycle of module voltage: Modulation II	86
7.39	Harmonic spectrum of module voltage: Modulation II	86
7.40	PWM operation of module voltage: Modulation II	87
7.41	Harmonic spectrum of module voltage at 15kHz: Modulation II	87
7.42	Experimental output voltage: Modulation I	88
7.43	Experimental output voltage: Modulation II	88
7.44	Harmonic spectrum of output voltage: Modulation I $\ \ldots \ \ldots \ \ldots \ \ldots$	89
7.45	Harmonic spectrum of output voltage: Modulation II	89
8.1	Converter Topology to Interface low voltage DC with three-phase Grid	90
8.2	Control Diagram for a PWM-CSI	95
8.3	Simulation Results	96

Chapter 1

Introduction

1.1 Conventional interface to integrate renewables to grid

Recent efforts to reduce the dependence on fossil fuels have caused utilities worldwide to increase the integration of the power generated from renewable energy systems [1]. This interconnection of renewable systems requires the use of grid tied inverters. A conventional utility interface utilizes two level inverters in conjunction with a line frequency transformer (LFT) to generate the required voltage as shown in Fig. 1.1. Such two level inverters have high total harmonic distortion (THD) in the generated output voltage and hence require the use of large filters. Also, the presence of common mode voltages with large $\frac{dv}{dt}$ require expensive common mode chokes or design of active filters [2]. Multilevel topologies with a single DC source and several LFTs to generate multilevel three-phase output voltage with low THD and reduced filter requirements have been proposed in literature [3]. However, in all of these topologies, the usage of LFTs increases the overall footprint of the system.



Figure 1.1: Conventional topology to interface low voltage DC to three-phase high voltage AC

1.2 Conventional interface for all electrified ships

With the advancements in variable speed drives, electrified ship propulsion systems as shown in Fig. 7.2 have become increasingly popular as they offer a compact size, increased fuel efficiency and reduced emissions in comparison to the mechanically propelled ships [4].



Figure 1.2: Electrified Ship Propulsion Systems.

These advantages have led to an increased interest in all electrified ships (AES) [5–9]. In an AES, the propulsion systems (power generation module, power conversion module, electric propulsion module) and ship load systems are unified to form an integrated system. With the availability of renewable energy resources and advancements in power electronics, more flexibility is available through an integrated system [9,10]. Recently, a lot of research has been done on different ways to integrate renewables and Battery Energy Storage Systems (BESS) to the upcoming electrified ships depending on the electrical distribution system [11–15]. The electrical distribution in a ship can either be completely DC, both DC and AC (hybrid), or it could be completely AC.

In ships proposed with MVDC distribution [11], the energy storage systems are integrated to the MVDC bus using a bi-directional DC-DC converter [16, 17]. Certain ships which employ hybrid distribution systems [12] the propulsion motors are connected to the AC distribution using a two stage AC-DC-AC converter (power converter shown in Fig. 1.1) with a DC-link capacitor. In these ships, it is proposed that the energy storage systems can be integrated to the DC-link capacitor directly, thereby eliminating an additional converter [18].

In some ships with AC distribution systems [13,19], the propulsion motors (synchronous) are connected directly to the distribution system, using a single stage converter. In such systems, the power converter (shown in Fig. 1.1) is either a thyristor controlled cycloconverter [17, 20] or any other direct AC-AC converter with bi-directional switches [21, 22]. The direct AC-AC conversion eliminates the DC-link capacitor, resulting in a more compact system with reduced maintenance [20]. In such ships, the low voltage energy storage systems require a Line Frequency Transformer (LFT) to interface with the high voltage AC distribution system(6.6 kV, 13.8 kV) [6, 17, 23]. The use of LFT in such a system counters the advantage that is attained by direct AC-AC conversion [23].



Figure 1.3: Proposed topology to interface low voltage DC to three-phase high voltage AC

1.3 Proposed interface using high-frequency transformers

At these power levels, high-frequency transformers (HFT) are envisioned to replace such LFTs [24, 25]. The use of HFTs reduces the size of magnetics but also necessitates a secondary side converter to generate line frequency AC from the high-frequency. This conversion can be accomplished either by a high-frequency AC-DC-line frequency AC, which is a two stage conversion process utilizing an electrolytic capacitor at the DC link [26–31] or a direct single-stage conversion from high-frequency AC-line frequency AC [32–38] as shown in Fig. 1.3. In case of the two stage conversion process, different topologies and modulation strategies have been investigated in the literature. In [26, 29, 30], the generated high frequency voltage across the transformer is converted to DC using a diode bridge and then phase shifted carriers are used to generate a multilevel voltage using series connected hard switched H-bridge modules. Dual active bridge topologies have been very common for isolated DC-DC conversion due to the inherent advantage of soft-switching. A review of various modulation strategies based on dual active bridge principle has been outlined in [39]. Recently, for isolated DC-DC application, instead of full bridge on two sides of the HFTs, MMCs (Modular Multilevel Converters) have also been investigated on one side [40, 41] and even on both sides [42]. Topologies have been proposed which extend the DAB principle to multi-port [43] systems, for DC-AC [27,44] and also AC-AC applications [45], [46]. Nevertheless, the DAB topologies suffer with low efficiency when operating at nonoptimal condition due to high circulating current and incomplete ZVS [39]. To overcome the limitations and hence improve the efficiency, different control techniques for operation under non-optimal conditions [47, 48] and topologies with additional switching stages have been proposed in literature [49]. In any case, to interface with high voltage line frequency AC, a number of converter modules, each connected to an isolated secondary winding are cascaded in series similar to the multilevel structure [26, 27, 32]. But such a system will require a large number of isolated secondaries and converter modules if lower voltage rating devices are used. With the advancement of high voltage switches, [28] the number of converter modules and the transformer secondaries can be reduced [50]. However, the absence of several series connected modules, increases the THD and will either require large filters or the design of more complex filters like LCL [51] or LLCL [52].

Asymmetrical multilevel inverters allow to achieve, higher number of levels in the output voltage with less number of converter modules. Asymmetrical converters can be broadly classified according to the source of asymmetry. In literature, topologies have been investigated in which the asymmetry is introduced in the DC voltage source [53–57]. Several unequal voltage sources have to be used to provide the required asymmetric input voltage. Recently, topologies have been proposed which have extended the asymmetric voltage configuration to modular multilevel converters [58]. Another scheme of generating asymmetric voltages is by using a single DC voltage source with a multi-winding transformer with unequal turns on each secondary. The asymmetry introduced in the LFT turns ratio have been previously presented [59,60]. However, as explained earlier, use of LFT increases the system size. Topologies with the asymmetry introduced in the high-frequency transformer turns ratio have been presented for drives [61]; traction [62] and for grid-tied applications [31,63–65]. In such two stage topologies, a high voltage DC input is still required for the main converter and the high-frequency transformer is used for auxiliary converters [61, 62]. But as stated, the two stage conversion requires a DC-link capacitor. An asymmetrical multilevel converter topology having nine secondary windings has been proposed, but the results do not incorporate switch non-idealities and its effect on the output voltage profile [63].

In this thesis, a multilevel converter topology to interface a low voltage DC to a threephase high voltage grid with three multi-winding HFTs is presented. The asymmetry is introduced in the HFT turns ratio (the three transformer secondary windings in each phase have an unequal turns ratio of 1:2:4). Such an asymmetry results in higher number of levels in the output voltage with only three series connected modules per phase. The single-stage conversion approach requires AC-AC converter modules. The proposed topology has the following benefits: (i) eliminates the bulky LFT; (ii) allows bi-directional power flow; (iii) generates fifteen output voltage levels, using only three AC-AC converter modules in each phase; and (iv) removes the need for a dc link capacitor, resulting in a reduced footprint. This converter can be used to integrate the energy storage system to HV AC distribution system as shown in Fig. 1.4.



Figure 1.4: High-frequency transformer isolated single stage topology to interface BESS to electrified ships.

1.4 Modulation and commutation strategies

Different modulation strategies have been investigated in the literature for cascaded Hbridge systems [66,67]. Phase shifted carriers are used to generate signals in high frequency link isolated single/two stage cascaded H-bridge systems to interface energy storage systems to the three-phase grid [26,27,68]. Primarily, two modulation techniques exists in the literature for asymmetrical cascaded H-bridge converter. These converters as mentioned above utilize the LFT turns ratio for generating the asymmetry and require a diode-bridge and a DC-link capacitor at the input of the H-bridge. Conventional hybrid modulation technique [69] and Uni-polar modulation technique [70]/hybrid frequency carrier based PWM [71] are the two techniques. The conventional hybrid modulation technique [69] allows to generate a multilevel output voltage with the dominant voltage harmonics at the sidebands of multiples of twice the switching frequency. However, the technique causes a rise of the DC-link capacitor voltage because of the negative DC-link current and requires a controlled rectifier at the input. The implementation of the uni-polar modulation rather than the conventional hybrid modulation technique in such a converter allows to get rid of the above disadvantages [70].

The single stage converter under study consists of AC-AC converter modules formed using bi-directional switches. The presence of bi-directional switches requires either a commutation strategy or a free-wheeling commutation path during each switching transition. As shown in the literature, the incorporation of a commutation strategy minimizes the losses in comparison to the freewheeling circuit [72]. Several strategies like the current based [73], voltage based [74] and hybrid [75] commutation have been investigated in the literature which can be classified as under:

- Current Based Commutation [73] strategy based on the output current sensing performs the commutation in four steps.
- Voltage Based Commutation [74] strategy based on input AC voltage sensing allows

the commutation in two steps.

• Hybrid Commutation [75] strategy requires both input AC voltage and output current sensing. It allows to reduce the commutation delay time because of single step commutation.

A detailed comparison of different current and voltage commutation strategies on the basis of efficiency and hardware requirement [76] show the superiority of four-step commutation strategy over other described strategies. However, the incorporation of four-step commutation algorithm can cause glitches in the voltage as shown in the literature [77, 78]. In literature, commutation strategies have been implemented to incorporate the non-idealities of the bi-directional switch in a single stage isolated two-level [79] and multi-level converter [80]. However, the non-ideal behavior of the switches of both the input H-bridge and the output single stage converter and the resulting effect on the output voltage profile has not been reported [32, 64, 79, 80].

A detailed description of various modes of circuit operation for a particular switching transition, highlighting the cause and effect of the glitch is demonstrated. The thesis proposes two compensation techniques to improve the output voltage profile, by modifying the commutation algorithm to negate the glitches in the output voltage. The switching of AC-AC converter module will result in square currents through the transformer winding [81] with a low frequency sinusoidal envelope. The presence of the transformer leakage inductance and square link currents require a clamp circuit in parallel to the transformer windings to avoid any over-voltage. However, the HFTs can be designed to achieve low leakage inductance [82–84]. The thesis also shows a winding design and characterization of a multi-winding transformer to achieve low leakage inductance.

The modified version of the uni-polar pulse width modulation technique is investigated for the single stage converter in [63]. However, the adoption of both the uni-polar modulation technique [71] and its modified version [64] to the subsequent converter topologies, result in voltage harmonics at the sidebands of the multiples of the switching frequency.

The single stage converter topology introduced in this thesis does not have an electrolytic capacitor at the DC-link. The absence of such a DC-link capacitor allows to investigate the hybrid modulation technique which has the aforementioned advantage. The hybrid modulation technique is modified based on the high-frequency link. The thesis investigates this modified modulation technique on a single stage isolated asymmetrical converter. Unlike previously used techniques, the proposed modulation shifts the dominant harmonics in the output voltage to the sidebands of multiples of twice the switching frequency, thereby reducing the output filter size. The thesis presents a detailed description for the generation of the ideal switch signals using the reference and carrier signals and also provides a qualitative

analysis to explain the harmonic spectrum of the output voltage.

The incorporation of the switch non-idealities requires the implementation of a commutation strategy. The thesis demonstrates the effect of the non-idealistic nature of the switch on the output voltage profile with a step-by-step modification in the switching signals as shown in the later sections. A detailed circuit description including the switch non-idealities to explain a particular switching transition is presented. The analysis explains the distortions in the output voltage as a result of switch non-idealities in comparison to the ideal output voltage. The analysis in general can be used for any isolated single stage converter and will also aid in the real-time implementation of the converter.

- Chapter 1 gives an introduction.
- Chapter 2 briefly presents the proposed circuit topology and the circuit description.
- In Chapter 3 the two investigated ideal modulations are presented along with a comparison based on output voltage THD and the harmonics.
- Chapter 4 explains the effect of switch non-idealities on the output voltage profile. Two glitch compensation techniques have been described in detail to improve the output voltage profile.
- In Chapter 5 a high-frequency four winding transformer design is detailed. Comprehensive analysis has been done to determine the transformer leakage inductances using 3-D Finite Element Analysis and using Network Analyzer.
- Chapter 6 demonstrates the operation of the proposed converter in case of a module failure. Also, a filter design comparison is outlined for the two different modulations.
- Chapter 7 shows the hardware setup and the detailed simulation and experimental results to validate the proposed concepts.
- Chapter 8 outlines two different circuit variations derived from the proposed topology and the basic simulation results.
- Chapter 9 gives the conclusion and the future work.

Chapter 2

High-Frequency Link Single-Stage Converter

2.1 Multilevel converter with high-frequency transformer (HFT) isolation

2.1.1 Circuit configuration

As shown in Fig. 2.1, in order to interface low voltage renewables having nominal dc voltage, V_{in} with a high voltage three-phase grid, full-bridge modules on the primary side of multi-winding HFTs of all three-phases are connected in parallel and on the secondary side AC-AC converter modules of each phase are connected in series. The number of AC-AC converter modules (N_m) required in each phase to interface with a three-phase grid, can be computed using (8.1). The voltage on the secondary side of the multi-winding transformer is $n_{si}V_{in}$, where, n_{si} : $i \in \{1, 2...N_m\}$ is the turns ratio of each isolated secondary winding to primary winding. N_m depends on v_{ll} , the rms line-line voltage of the three-phase grid and nV_{in} , the voltage across the secondary winding of the HFT. Considering, close to 50% de-rating of the switches, v_r the rating of the IGBTs that will be used to design the AC-AC converter module is given by (8.3). N_m , series connected modules in each phase will result in a phase-neutral output voltage having n_l number of levels as given by (2.3). Each AC-AC converter module consists of eight IGBTs, and hence, the total number of IGBTs in each phase is N_{sw} and can be computed using (2.4).

$$N_m = \frac{\sqrt{2}v_{ll}}{\sqrt{3}nV_{in}} \tag{2.1}$$

$$v_r = \frac{nV_{in}}{0.5} \tag{2.2}$$

$$n_l = 2N_m + 1 \tag{2.3}$$

$$N_{sw} = 8N_m \tag{2.4}$$

Typically, in a cascaded system, all the converter modules are symmetrical. Suppose, Fig. 2.1 is considered to be a symmetrical converter topology, then all the transformer secondary windings, $n_{s1} = n_{s2} = ...n_{sN_m} = n$ will have identical turns ratio. Also, the IGBTs in all the series connected AC-AC converter modules will have the same voltage rating. However, it is also possible to design the multi-winding transformer such that all the isolated secondary windings have unequal turns ratio; $n_{s1} \neq n_{s2} \neq ...n_{sN_m}$. This will result in higher number of output voltage levels with less number of converter modules. However, each of the AC-AC converter module needs to have IGBTs rated at different voltages.

2.1.2 Evaluation of circuit configuration based on system complexity

Table 2.1 shows a comparison of the number of IGBTs and transformer secondary windings required to realize the grid voltage based on the voltage rating of the IGBT. The number of AC-AC converter modules to be connected in series, the number of output voltage levels and the total number of IGBTs in each phase are calculated using (8.1), (2.3) and (2.4)respectively and shown in Table 2.1. In Table 2.1, a refers to the case in which the grid voltage is realized using 6.5 kV IGBTs. This results in only five levels in the output voltage resulting in a high output voltage THD. As shown in case b and c, with the symmetrical modules, the number of output voltage levels can be increased by selecting lower voltage rating IGBTs. However, this will increase the number of IGBTs and the transformer secondary windings as shown in Table 2.1. Table 2.1 case d, is of the proposed asymmetrical converter topology in which the turns ratio of the secondary windings are asymmetrical. Several configurations of single-stage symmetrical and asymmetrical multilevel converters (7-S: 7-level symmetrical converter; 7-AS: 7-level asymmetrical converter and so on) have been compared based on system complexity as shown in Table 2.2. The number of switches along with the associated isolated gate-drivers and digital signals, the number of isolated secondary windings, the number of arithmetic and logical operations to generate the ideal switch signals and the resulting THD of a three-phase system have been summarized in Table 2.2. Based on the data, a normalized index is computed using [26] and shown in Table 2.3. As seen from Table 2.2, in comparison to 15- AS configuration only 7-S and 7-AS configurations have lower control complexity. However, the THD of a seven level converter is much higher. The normalized index in Table 2.3, shows that the control complexity (based on ALOs) in particular in asymmetrical fifteen level configuration though not the minimum, but in terms of normalized index and overall system performance is superior compared to other configurations.



Figure 2.1: Block diagram of proposed topology.

Case	Output Voltage	Switch Rating	Isolated Secondary	N_{sw}
a.	5	$6.5 \ kV$	2	16
b.	9	$3.5 \ kV$	4	32
с.	15	$1.7 \ kV$	7	56
		$6.5 \ kV$	1	8
d.	15	$3.5 \ kV$	1	8
		$1.7 \ kV$	1	8

Table 2.1: Switch count for 500 kW/7.2 kV interface

Levels	Switches	THD	ALO	Sec. Windings
7- S	84	19.61	114	3
7- AS	60	19.61	90	2
9- S	108	14.81	150	4
11- S	132	12.02	186	5
15- S	180	8.62	258	7
15- AS	84	8.62	150	3
17- S	204	7.59	294	8
27- AS	108	4.66	210	4

Table 2.2: System Complexity

Table 2.3: Normalized index value

Levels	Switches	THD	ALO	Sec. Windings	Total
7- S	0.17	1	0.12	0.17	1.46
7- AS	0	1	0	0	1
9- S	0.33	0.68	0.29	0.33	1.63
11- S	0.5	0.49	0.47	0.5	1.96
15- S	0.83	0.26	0.82	0.83	2.74
15- AS	0.17	0.26	0.29	0.17	0.89
17- S	1	0.2	1	1	3.2
27- AS	0.33	0	0.59	0.33	1.25



Figure 2.2: Ideal output voltage: (a) seven level;(b) fifteen level; (c)twenty seven level; Harmonic spectrum of ideal output voltage: (d) seven level;(e) fifteen level; (f)twenty seven level



Figure 2.3: Comparison of different configurations based on normalized index parameters for: control complexity, number of switches, transformer secondary windings and THD



2.1.3 Description of the circuit configuration

Figure 2.4: Detailed per-phase circuit diagram of isolated asymmetrical multilevel converter topology.

Fig. 2.4 shows the per phase circuit configuration of the proposed converter topology. In each phase i : a, b, c, the full-bridge module consists of switches P_{i1} - P_{i4} . The output of the full-bridge is connected to the primary winding of a HFT having three secondary windings with an asymmetrical turns ratio of n:2n:4n as shown in Fig. 2.4 (highlighted in red). Such an asymmetry, results in fifteen levels in the output voltage, same as in Table 2.1 case c, but with only three series connected AC-AC converter modules and three transformer secondaries per phase. Each AC-AC converter module is connected to an isolated transformer secondary winding. As shown in Table 2.1, case d and Fig. 2.4, the AC-AC converter Module I should have IGBTs rated at 6.5 kV, whereas Module II should have IGBTs rated at 3.5 kV and Module III at 1.7 kV. Each module consists of four bidirectional switches $S_{1a}-S_{4a}$. The bi-directional switch, S_{1a} can be realized by connecting two anti-parallel IGBTs (S_{1a1} and S_{1a2}) in a common emitter configuration as shown in Fig. 2.4. In each phase, all the AC-AC converter modules are connected in series on the output side. The module output voltage of Modules I, II and III are v_{a1} , v_{a2} and v_{a3} respectively. The net phase-neutral output voltage of the three series connected modules is v_a . The square currents through the leakage inductance of the transformer due to the switching of AC-AC converter module requires a clamp circuit. The circuit comprises of a diode full-bridge connected in parallel to the transformer windings to avoid any over-voltage.

Chapter 3

Ideal Modulation

3.1 Modulation-I: Modified Unipolar Modulation

3.1.1 Modulation of primary side full-bridge converter

The primary side full-bridge consists of four unidirectional switches $P_{a1} - P_{a4}$. In each leg, the two switches are regulated in a complementary manner to avoid the short-circuit of input dc voltage. The switches are regulated at 50% to generate a square voltage across the transformer primary winding. Switches P_{a1} and P_{a4} are controlled by signal, p_1 and the other two switches are controlled by signal p_2 . The switching strategy leads to a generation of $+V_{in}$ (when signal p_1 is high) and $-V_{in}$ (when signal p_2 is high) voltage at the transformer primary winding terminal (v_p) . Hence, the voltages across the three isolated secondary windings will be as given in (3.1), (3.2) and (3.3) respectively.

$$v_{sa1} = \pm 4nV_{in} \tag{3.1}$$

$$v_{sa2} = \pm 2n V_{in} \tag{3.2}$$

$$v_{sa3} = \pm n V_{in} \tag{3.3}$$

3.1.2 Modulation of secondary side AC-AC converter modules

The three AC-AC converter modules connected to each of the three isolated secondary windings, have to be controlled, so as to generate the multilevel phase-neutral output voltage from the link voltage that is present across the terminals of each secondary winding. The ideal modulation strategy for the proposed converter topology for phase, a is explained. Same modulation strategy will apply for phase, b and c. Fig. 3.1a shows the implementation

circuit for the generation of the ideal signals of the switches of the AC-AC converter module. Two complementary sinusoidal references are compared with seven level shifted carriers to obtain seven digital signals, $d_1 - d_7$, as in a uni-polar modulation [71]. The carrier waveform used to generate the ideal digital signals, is at the link frequency. As shown in Fig. 3.1a, three ideal signals, $p_{1a} - p_{3a}$ are generated from the logical operation of seven digital signals. Similarly, from the digital signals obtained by comparing the complementary sinusoidal reference with the level shifted carriers as shown in Fig. 3.1a, three more signals $n_{1a} - n_{3a}$ can be generated. In order to generate positive module output voltages, when the link voltage, v_{pa} is positive (that is, when signal p_1 is high), p_{1a} , p_{2a} and p_{3a} are q_{1a} , q_{5a} and q_{9a} signals respectively and n_{1a} , n_{2a} and n_{3a} are q_{2a} , q_{6a} and q_{10a} signals respectively as shown in Fig. 3.1a. When the link voltage turns negative (that is, when signal p_1 is low), the signals are interchanged to generate the required output voltage. The modulation is similar to the uni-polar modulation technique [71] except that the signals are modified based on the high-frequency link polarity. The modified signals are then given to the switches in the three AC-AC converter modules in each phase.

$$p_{1a} = d_4 \tag{3.4}$$

$$p_{2a} = d_2 \oplus d_4 \oplus d_6 \tag{3.5}$$

$$p_{3a} = d_1 \oplus d_3 \oplus d_5 \oplus d_7 \oplus p_{2a} \tag{3.6}$$



Figure 3.1: (a) Generation of ideal modified signals; (b) Switching signals of primary side full-bridge and AC-AC converter Module I followed by the phase-neutral output voltage, v_a : Ideal signals without four-step commutation of secondary side bi-directional switches and no dead time of primary side switches.

Module I, v_{a1}	Module II, v_{a2}	Module III, v_{a3}	Output Voltage, v_a
$4nV_{in}$	$2nV_{in}$	$nV_{in} \leftrightarrow 0$	$7nV_{in} \leftrightarrow 6nV_{in}$
$4nV_{in}$	$2nV_{in} \leftrightarrow 0$	$0 \leftrightarrow nV_{in}$	$6nV_{in} \leftrightarrow 5nV_{in}$
$4nV_{in}$	0	$nV_{in} \leftrightarrow 0$	$5nV_{in} \leftrightarrow 4nV_{in}$
$4nV_{in} \leftrightarrow 0$	$0 \leftrightarrow 2nV_{in}$	$0 \leftrightarrow nV_{in}$	$4nV_{in} \leftrightarrow 3nV_{in}$
0	$2nV_{in}$	$nV_{in} \leftrightarrow 0$	$3nV_{in} \leftrightarrow 2nV_{in}$
0	$2nV_{in} \leftrightarrow 0$	$0 \leftrightarrow nV_{in}$	$2nV_{in} \leftrightarrow nV_{in}$
0	0	$nV_{in} \leftrightarrow 0$	$nV_{in} \leftrightarrow 0$

3.1.3 Generation of specific output voltage level

Table 3.1: Module and total phase-to-neutral output voltages

Table 8.1 shows the output voltage across each of the three AC-AC converter module to generate seven positive levels in the phase-neutral output voltage. As shown in Table 8.1, for a particular output voltage level generation, if the module output voltage is constant then it will switch only when the the link voltage switches otherwise it will switch even

though the link voltage is constant. Fig. 3.2 shows the output voltage across Modules I, II and III followed by the total phase-neutral voltage, v_a depicting the fifteen levels, with three AC-AC converter modules in each phase. To illustrate with an example for phase a, consider the generation of the output voltage $7nV_{in} \leftrightarrow 6nV_{in}$. Fig. 3.1b shows the switching signals of the primary side full-bridge switches followed by the ideal signals of the bi-directional switches in Module I. The signals q_{1a} - q_{4a} correspond to bi-directional switches S_{1a} - S_{4a} respectively. In case of ideal switches, both uni-directional switches S_{1a1} and S_{1a2} forming a bi-directional switch, S_{1a} can be controlled using one signal, q_{1a} . As shown in Table 8.1, irrespective of the link voltage being positive or negative, Modules I and II have to generate a constant voltage of $4nV_{in}$ and $2nV_{in}$ respectively and Module III has to generate a voltage that is pulsating between nV_{in} and zero. The left leg switches of the AC-AC converter module are controlled by p_1 signal whereas, the right leg switches are controlled by p_2 . This implies that as shown in Fig. 3.1b, when the link voltage switches from negative (signal p_2 is high) to positive (signal p_1 is high), the bi-directional switches that are conducting, transition from $(S_{2a} \text{ and } S_{3a})$ in Module I; and $(S_{6a} \text{ and } S_{7a})$ in Module II to $(S_{1a} \text{ and } S_{4a})$ in Module I; and $(S_{5a} \text{ and } S_{8a})$ in Module II to maintain a constant module output voltage of $4nV_{in}$ and $2nV_{in}$ respectively. Switching in Module III will be similar to obtain nV_{in} . But Module III also generates a zero voltage across the output terminals as shown in Table 8.1. For this, both the bottom bi-directional switches will be ON to provide a path for the output current to free-wheel, irrespective of the link voltage being positive or negative. In a similar manner, the switches are controlled to achieve other voltage levels.



Figure 3.2: Voltage waveforms for case 3.1b for 1 pu V_{in} and n=1

3.2 Modulation-II: Modified Hybrid Modulation

3.2.1 Modulation of primary side converter

The ideal modulation technique for phase, a is described in this section. Same applies for phase, b and c. In the primary side H-bridge, each diagonal pair of switches is regulated in a complementary manner at 50 % duty ratio. Diagonal pair of switches P_{a1} and P_{a4} is controlled by signal p_1 and the other diagonal pair of switches is controlled by p_2 resulting in a following link voltage, v_{pa} across the transformer primary winding:

$$v_{pa} = \begin{cases} +V_{in} & p_1 = 1, p_2 = 0\\ -V_{in} & p_1 = 0, p_2 = 1 \end{cases}$$
(3.7)

$$m_a = \frac{\hat{V}_{a,1}}{7nV_{in}} \tag{3.8}$$

where, $\hat{V}_{a,1}$ is the peak value of the fundamental output voltage.

3.2.2 Signal generation for switches of secondary side modules

The voltages across the three isolated secondary windings, v_{sa1} , v_{sa2} and v_{sa3} will be $4n v_{pa}$, $2n v_{pa}$ and $n v_{pa}$ respectively. The three AC-AC converter modules connected to each of

the three isolated secondary windings, are modulated, so as to generate a multilevel phaseto-neutral output voltage. The modulation index, m_a is given in (7.2). Table 3.2 shows the output voltage across each of the three AC-AC converter modules to generate a multilevel output voltage. For simplicity, the switches of the AC-AC converter modules are considered to be ideal. Similar to the primary side H-bridge modulation, two bi-directional switches of each leg are controlled in a complementary fashion to avoid shorting of the transformer secondary winding.

Module I	Module II	Module III	Output Voltage
$4nV_{in}$	$2nV_{in}$	$nV_{in} \leftrightarrow 0$	$7nV_{in} \leftrightarrow 6nV_{in}$
$4nV_{in}$	$2nV_{in}$	$0 \leftrightarrow -nV_{in}$	$6nV_{in} \leftrightarrow 5nV_{in}$
$4nV_{in}$	$2nV_{in} \leftrightarrow 0$	$-nV_{in} \leftrightarrow 0$	$5nV_{in} \leftrightarrow 4nV_{in}$
$4nV_{in}$	$0 \leftrightarrow -2nV_{in}$	$0 \leftrightarrow nV_{in}$	$4nV_{in} \leftrightarrow 3nV_{in}$
$4nV_{in}$	$-2nV_{in}$	$nV_{in} \leftrightarrow 0$	$3nV_{in} \leftrightarrow 2nV_{in}$
$4nV_{in}$	$-2nV_{in}$	$0 \leftrightarrow -nV_{in}$	$2nV_{in} \leftrightarrow nV_{in}$
$4nV_{in} \leftrightarrow 0$	$-2nV_{in} \leftrightarrow 0$	$-nV_{in} \leftrightarrow 0$	$nV_{in} \leftrightarrow 0$

Table 3.2: Modulation Technique for Different Positive Levels in Phase-to-Neutral Output Voltage



Figure 3.3: (a) Generation of ideal modified signals; (b) Switching signals of primary side full-bridge and AC-AC converter Module I followed by the phase-neutral output voltage, v_a : Ideal signals without four-step commutation of secondary side bi-directional switches and no dead time of primary side switches.

Hence, ideally each leg of the AC-AC converter module can be controlled with one switching signal. Signals s_{1a} and s_{2a} are for the top bi-directional switch in left and right leg respectively of Module I (shown in Fig. 2.1). Similarly, s_{5a} and s_{6a} are signals for the switches of Module II and s_{9a} and s_{10} of Module III. The ideal switch signals are generated using seven level shifted carriers and two complementary sinusoidal reference signals as shown in Fig. 3.4a. In order to generate seven positive levels in the output voltage these carriers are compared with a sinusoidal reference as shown in Fig. 3.3a; such that whenever the sinusoidal reference is greater than the carrier the logic is 1, otherwise 0. As shown in Fig. 3.3a, three ideal signals, $p_{1a} - p_{3a}$ are generated from the logical operation of seven digital signals as under:



Figure 3.4: (a) Seven carrier signals and two complementary references to generate fourteen digital signals.(b) Three ideal signals derived from seven digital signals to generate fifteen levels using three modules.

These three signals are shown in Fig. 3.4b. Similarly, as shown in Fig. 3.3a, three more signals $n_{1a} - n_{3a}$ can be generated from seven other digital signals obtained by comparing the complementary sinusoidal reference with the level shifted carriers. In order to generate positive module output voltages, when the link voltage, v_{pa} is positive (that is, when signal p_1 is high), p_{1a} , p_{2a} and p_{3a} are s_{1a} , s_{5a} and s_{9a} signals respectively and n_{1a} , n_{2a} and n_{3a} are s_{2a} , s_{6a} and s_{10a} signals respectively as shown in Fig. 3.3a. When the link voltage turns negative (that is, when signal p_1 is low), the signals are interchanged to generate the required output voltage. It is to be noted that, if according to Table 3.2, the module has to generate a constant voltage for a specific output voltage level, v_a , it will switch only when the link voltage changes polarity. However, if the module has to generate a pulsating voltage (signal p_{3a}), then as shown in Fig. 3.4b it will switch even though the link voltage is constant.

3.2.3 Generation of specific output voltage level

To illustrate with an example for phase a, consider the generation of the output voltage $6nV_{in} \leftrightarrow 5nV_{in}$. As shown in Table 3.2, irrespective of the link voltage polarity, Modules I and II have to generate a constant voltage of $4nV_{in}$ and $2nV_{in}$ respectively, and Module III has to generate a voltage that is pulsating between 0 and $-nV_{in}$. Fig. 3.3b shows the ideal switching signals of the primary side H-bridge switches, p_1 and p_2 followed by the signals of the two bi-directional switches in the left leg: $(s_{1a} \text{ and } s_{3a})$ of Module I and $(s_{9a} \text{ and } s_{11a})$ of Module III in order to generate $v_a=5nV_{in}$. Fig. 4.13 shows a detailed circuit diagram highlighting different modes of operation in order to maintain an output voltage of $5nV_{in}$ even though the link voltage changes from $+V_{in}$ to $-V_{in}$. As modules I and II switch

only when the input H-bridge converter changes state, the signals $(s_{1a} \text{ and } s_{5a})$ of the left leg top bi-directional switch and signals $(s_{2a} \text{ and } s_{6a})$ of the right leg top bi-directional switch of both modules follow signals p_1 and p_2 respectively. Mode 1 of Fig. 3.3b shows that signal p_1 is high, corresponding to a link voltage, $v_{pa}=+V_{in}$ as explained earlier. Mode 1 of Fig. 4.13a shows the corresponding bi-directional switch set in both modules that are ON resulting in module output voltages of $4nV_{in}$ and $2nV_{in}$ respectively. As Module III needs to generate a voltage of $-nV_{in}$, the switching signals will be complimentary as shown in Fig. 3.3b. Similarly, as shown in Fig. 4.12a, when the signal p_2 is high, the corresponding bi-directional switch set in all three modules are ON, such that the phase-toneutral output voltage is still $5nV_{in}$. In order to generate $6nV_{in}$, Module III has to generate a zero voltage. Whenever a zero voltage is to be generated, two bi-directional switches of both legs in Module III: S_{11a} and S_{12a} are ON to provide a free-wheeling path to the current irrespective of the link voltage being positive or negative. In a similar manner, the switches are controlled to achieve other levels.



Figure 3.5: Output voltages, v_{a1} , v_{a2} and v_{a3} across each of the three Modules I, II and III respectively, followed by the total phase-to-neutral output voltage, v_a , with a switching frequency of 15 kHz using the proposed modulation

3.3 Harmonic spectrum of voltage for both modulation techniques

3.3.1 Spectrum of module output voltage: Modulation-I

Fig. 3.6 shows the module output voltages and the harmonic spectrum for the modified uni-polar modulation technique. As shown in Fig. 3.6a the leg voltage, v_{l1} generates both positive and zero output voltage during four of the seven positive output voltage levels in a fundamental cycle. As the modulation technique is implemented on the converter topology proposed in [63], the right leg also generates both negative and zero voltage during four of the seven positive output voltage levels. This is different as compared to the conventional modulation techniques with no zero voltages. However, as shown in Fig. 3.6b, the extra switching of the right leg to generate the positive output voltage level still has dominant voltage harmonics at the switching frequency and its sidebands not only for the two leg voltages but also for the module voltage, v_{a1} . Fig. 3.6c shows the two leg voltages and the module output voltage, v_{a1} during the PWM operation. As it can be clearly seen the right leg of the module does generate a negative voltage but as shown in Fig. 3.6d, the harmonics remain at the sidebands of 15kHz.



Figure 3.6: Leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1} using modified uni-polar modulation technique (a) Fundamental cycle; (b) Harmonic spectrum ; (c) zoomed in to show the voltage profile during PWM operation; (d) Harmonic spectrum at 15kHz to show that the switching frequency harmonic component is not canceled in the module output voltage, v_{a1} .
3.3.2 Spectrum of module output voltage: Modulation-II

Fig. 3.7 shows the module output voltages and the harmonic spectrum for the modified hybrid modulation technique. As shown in Fig. 3.7a the leg voltage, v_{l1} generates both positive and zero output voltage throughout the seven positive output voltage levels in a fundamental cycle. As the modulation technique is implemented on the converter topology, the right leg also generates both negative and zero voltage during all the seven positive output voltage levels. As shown in Fig. 3.7b, the output voltage has dominant voltage harmonics at the sidebands of twice the switching frequency although the two leg voltages have dominant harmonic at the switching frequency and its sidebands. Fig. 3.7c shows the two leg voltages and the module output voltage, v_{a1} during the PWM operation. As it can be clearly seen both the left and the right leg of the module switch and this results in the output voltage having multiple pulses in a 15kHz interval. Hence, as shown in Fig. 3.7d, the harmonics cancel out at 15kHz and its sidebands.

3.3.3 Spectrum of output voltage

Fig. 3.8a and Fig. 3.8b shows the output voltage across each of the three modules followed by total voltage, v_a using the proposed modulation and the modified uni-polar modulation technique by considering ideal switches of both the primary side H-bridge and the AC-AC converter modules. Fig. 3.8c shows the harmonic spectrum of the output voltage, v_a shown in Fig. 3.8a. As shown in the Fig. 3.8c, the dominant harmonics are at 30kHz which is twice the switching frequency. Fig. 3.8d shows the harmonic spectrum of the output voltage, v_a shown in Fig. 3.8b. As shown in the Fig. 3.8d, the dominant harmonics are at the switching frequency of 15kHz. The harmonic spectrum of the output voltage using both the techniques show that, the proposed modulation allows to achieve the dominant harmonics at twice the switching frequency which leads to a smaller grid filter inductance.



Figure 3.7: (a)Leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1} ; (b) Harmonic spectrum of the leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1} ; (c)Leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1} zoomed to show the switching during PWM operation; (d) Harmonic spectrum of the three voltages at 15kHz to show the cancellation of the switching frequency harmonic component in the module output voltage, v_{a1} .



Figure 3.8: Output voltages, v_{a1} , v_{a2} and v_{a3} across each of the three Modules I, II and III respectively, followed by the total phase-to-neutral output voltage, v_a , with a switching frequency of 15 kHz. (a) Proposed Modulation (b) Modified uni-polar modulation presented in [63]. Harmonic spectrum of the output voltage with a switching frequency of 15 kHz(c) Proposed Modulation (d) Modified uni-polar modulation technique

Chapter 4

Non-Ideal Modulation

4.1 Modulation-I: Incorporation of Four-step commutation and Dead-time

4.1.1 Effect on output voltage profile

The previous chapter described the process of the ideal switching from the bottom bidirectional switch S_{3a} to S_{1a} in the left leg of Module I when the link switches from $-V_{in}$ to $+V_{in}$ in order to maintain constant $7nV_{in}$ output voltage. In this section, the effect of switch non-idealities on the phase-neutral output voltage is investigated. The primary side full-bridge switches are controlled using signals a and b with a dead-time interval. On the secondary side each leg of AC-AC converter module follows a four-step commutation algorithm, so that the continuous output current is not interrupted and the isolated secondary windings are not short-circuited. Hence, both uni-directional switches S_{1a1} and S_{1a2} forming a bi-directional switch are controlled independently using q_{1a1} and q_{1a2} respectively. For this case, the same switching transition occurs in all AC-AC converter modules. Hence, the switching transition is described by considering only the primary side full-bridge module and Module I of phase-a. Fig. 4.1a shows the switching signals of the primary side switches and the bi-directional switches of Module I followed by the phase-neutral output voltage for that switching transition. The transition will occur in six modes of operation depicted below as mode I to mode VI in Fig. 4.2: blue indicates the gate signal and red indicates the current path. The output current direction is as shown in Fig. 4.2, and assumed to be positive. A detailed description of different modes is explained below:



Figure 4.1: Switching signals of primary side full-bridge and AC-AC converter Module I followed by the phase-neutral output voltage, v_a : (a) with four-step commutation of secondary side bi-directional switches and dead time of primary side switches (b) with four-step commutation of secondary side bi-directional switches and dead time of primary side switches along with glitch compensation-I (c) with four-step commutation of secondary side bidirectional switches and dead time of primary side switches along with glitch compensation-I for reverse power flow



Figure 4.2: Circuit diagram for the various modes of opearation for the case 4.1a for 1 pu input voltage, V_{in} and n=1

- (1) Mode I: In this mode, as shown in Fig. 4.1a, b is high and a is low. Mode I of Fig. 4.2 shows the circuit representation in which the switches P_{a2} and P_{a3} are ON. This will result in a negative voltage $-4nV_{in}$ across the link, on the transformer secondary side, as indicated by the arrow pointing downwards. As shown in Fig. 4.1a signals q_{2a1} and q_{2a2} ; q_{3a1} and q_{3a2} are high. This corresponds to the bi-directional switches S_{2a} and S_{3a} being ON, as shown in mode I of Fig. 4.2 resulting in a positive module output voltage, $v_{a1} = 4nV_{in}$. Similarly, corresponding switches in Modules II and III will also conduct such that, $v_{a2} = 2nV_{in}$; $v_{a3} = nV_{in}$ and the total phase-neutral output voltage, v_a is $7nV_{in}$ as shown in 4.1a.
- 2) Mode II: At the beginning of mode II, as shown in Fig. 4.1a, signal b is gated low. Corresponding to that, as explained in ideal modulation section, signal q_{2a} goes low.

Hence, the top bi-directional switch of the right leg undergoes the first step of fourstep commutation (FSC): the non-conducting switch (S_{2a1}) of the outgoing switch pair (S_{2a}) is gated low (q_{2a1}) . Also, in this mode, signal *a* and hence, the ideal signal, q_{1a} of left leg, does not undergo any switching transition. As shown in mode *II* of Fig. 4.2, the link current, i_{sa1} is negative. On the transformer primary side, as the link current is negative and all switches are turned OFF, as shown in mode *II* of Fig. 4.2 diodes of the switches P_{a1} and P_{a4} conduct, and this results in positive link voltage $+4nV_{in}$, shown by the arrow pointing upwards. For the positive link voltage, as the switch-diode pair explained above on the secondary side conduct, v_a voltage of, $-7nV_{in}$, as shown in mode *II* of Fig. 4.1a is generated.

- 3) Mode III: Mode III of Fig. 4.2 shows the circuit representation in which the switches P_{a1} and P_{a4} are ON, as signal *a* goes high. This will result in $+4nV_{in}$ across the link on the secondary side, as shown by the arrow pointing upwards. On the secondary side, as signal *a* undergoes transition hence, the ideal signal, q_{3a} goes low. Hence, in the left leg the first step of FSC occurs: the non-conducting switch (S_{3a1}) of the outgoing switch pair (S_{3a}) is gated OFF (q_{3a1}) . The current continues to flow through S_{3a2} and D_{3a1} pair. In case of the right leg switch pair, the second step of FSC occurs: conducting switch (S_{4a1}) of the incoming switch pair (S_{4a}) is gated ON (q_{4a1}) . In the right leg, even though S_{2a2} is still ON, as the link voltage is positive, the diode D_{2a1} is reverse biased. Hence, the current starts to flow through the S_{4a1} and D_{4a2} pair. This results in a zero voltage, v_a as shown in mode III of Fig. 4.1a.
- 4) Mode IV: In this mode, $+4nV_{in}$ voltage across the link on the secondary side is maintained. As shown in mode IV of Fig. 4.2, in the left leg of the secondary side module, second step of FSC occurs: the conducting switch (S_{1a1}) of the incoming switch pair (S_{1a}) is gated ON (q_{1a1}) . In the left leg, even though S_{3a2} is still ON, as the link voltage is positive, the diode D_{3a1} is reverse biased. Hence, the current flows through the S_{1a1} and D_{1a2} pair, as D_{1a2} is forward biased. In case of the right leg switch pair, the third step of FSC occurs: the conducting switch (S_{2a2}) of the outgoing switch pair (S_{2a}) is gated OFF (q_{2a2}) and the current continues to flow through S_{4a1} and D_{4a2} pair as in mode III. $7nV_{in}$ output voltage, is generated as shown in mode IV of Fig. 4.1a.
- 5) Mode V: For $+4nV_{in}$ across the link on the secondary side, as shown in Fig. 4.1a, in the left leg, the third step of FSC occurs: the conducting switch (S_{3a2}) of the outgoing switch pair (S_{3a}) in the left leg is gated OFF (q_{3a2}) . The current continues to flow through the S_{1a1} and D_{1a2} pair. In case of the right leg switch pair, the last step of

the FSC takes place. The non-conducting switch (S_{4a2}) of the incoming switch pair (S_{4a}) is gated ON (q_{4a2}) . This is represented as the bi-directional switch S_{4a} being turned ON as shown in mode V of Fig4.2. $7nV_{in}$ output voltage is maintained, as shown in mode V of Fig. 4.1a.

6) Mode VI: In this mode, for $+4nV_{in}$ across the link on the secondary side, as shown in Fig. 4.1a, in the left leg the last step of FSC occurs: the non-conducting switch (S_{1a2}) of the incoming switch pair (S_{1a}) is gated ON (q_{1a2}) . This is represented as the bi-directional switch S_{1a} is turned ON.



Figure 4.3: Circuit diagram for the various modes of operation for the case 4.1b

The different modes of operation as explained above show that, contrary to the ideal case wherein the output voltage, v_a is always positive $7nV_{in}$ as shown in Fig. 3.2, the output voltage is $-7nV_{in}$ in mode II and zero in mode III as shown in Fig. 4.1a. Fig. 4.4 shows the output voltage across Modules I, II and III followed by, v_a with switch non-idealities. The generated output voltage has a voltage profile different as compared to the ideal modulation strategy as highlighted by the pink and blue patches in the output voltage, v_a .

4.1.2 Glitch Compensation-I

In this section, a compensation technique is proposed to negate the negative voltage glitch which occurs in mode II of Fig. 4.1a. Fig. 4.3 shows the different modes of circuit operation for the exact same transition as in the previous section but with glitch compensation. Modes v and vi of Glitch Compensation-I are same as modes V and VI, and hence, not shown in Fig. 4.3. A detailed description of different modes of operation is explained below:

- 1) *Mode i*: In this mode, as shown in Fig. 4.1b and Fig. 4.3, the switching signals and the circuit operation is exactly the same as mode *I*.
- 2) Mode *ii*: In this mode, as shown in Fig. 4.1b, b^* is still ON. This mode of operation is different from mode *II* of Fig. 4.1a and Fig. 4.2 where *b* was gated OFF. The secondary side switching is still determined by signals *a* and *b*. Hence, it remains the same as that of mode *II*. As shown in mode *ii* of Fig. 4.3, as the switches P_{a2} and P_{a3} still conduct, this maintains the link voltage on the secondary side as $-4nV_{in}$, as shown by the arrow pointing downwards. For the negative link voltage, the switch-diode pair as explained above on the secondary side conduct, resulting in $7nV_{in}$ voltage, v_a as shown in mode *ii* of Fig. 4.1b, hence, negating the negative voltage $-7nV_{in}$ as shown in mode *II* of Fig. 4.1a.
- 3) Mode iii: In this mode, as shown in Fig. 4.1b, b* is gated OFF. All the switches of the primary side are gated OFF. On the secondary side, as explained in mode III, output current freewheels through the bottom two switch-diode pair resulting in zero voltage across the link and the output as shown in mode III of Fig. 4.1b.
- 4) Mode iv: In this mode, as shown in Fig. 4.1b, a* is gated ON. As shown in mode iv of Fig. 4.3 this results in a positive voltage +4nV_{in} across the link. For the positive link voltage as explained above in mode IV, the corresponding switch-diode pair on the secondary side conduct, resulting in 7nV_{in} voltage, v_a as shown in mode iv of Fig. 4.1b.
- 5) Mode v and vi: The modes of operation are the same as in modes V and VI.

The different modes of operation as explained above, show that the glitch compensation allows to negate the $-7nV_{in}$ voltage occurring in mode II of Fig. 4.1a, as shown in mode iiof Fig. 4.1b. Fig. 4.5 shows the output voltage across Modules I, II and III followed by, v_a with switch non-idealities. The output voltage does not go to $-7nV_{in}$ but still goes to zero in each switching cycle. Along with the distortion caused during the link voltage transition, the output voltage v_a , also has distortion during the generation of certain voltage levels when the link is constant as shown in Fig. 4.4 (highlighted by blue patches) which are not negated by glitch compensation-I as shown in Fig. 4.5.



Figure 4.4: Voltage waveforms of series connected converter modules followed by phaseneutral voltage for a fundamental cycle for 1 pu input voltage, V_{in} and n=1: for case Fig. 4.1a without any compensation.



Figure 4.5: Voltage waveforms of series connected converter modules followed by phaseneutral voltage for a fundamental cycle for 1 pu input voltage, V_{in} and n=1: for Fig. 4.1b with Glitch Compensation-I.



Figure 4.6: Voltage waveforms of series connected converter modules followed by phaseneutral voltage for a fundamental cycle for 1 pu input voltage, V_{in} and n=1: for case 4.1c with glitch compensation-I and reverse power flow.



Figure 4.7: Voltage waveforms of series connected converter modules followed by phaseneutral voltage for a fundamental cycle for 1 pu input voltage, V_{in} and n=1: with Glitch Compensation I and II.

4.1.3 Glitch Compensation-I (during reverse power flow)

The modulation and commutation pertaining to the power flow in the reverse direction that is from the three-phase grid to the DC supply is shown in Fig. 4.1c. It is to be noted that the primary side signals and the secondary side ideal switch signals will remain the same but the commutation will change as the current direction is reversed. Fig. 4.1c and 4.8 show the switching signals and the modes of circuit operation respectively for the exact same transition but with the current direction reversed. Based on the current direction, as shown in Fig. 4.1c, the corresponding switches conduct on the secondary side converter during the various modes of operation. The resulting output voltage after glitch compensation-I is $7nV_{in}$ in modes a, b, e and f and zero in c. This is similar to Fig. 4.1b for positive current. However, in mode d, the operation is different as shown below:



Figure 4.8: Circuit diagram for the various modes of opearation for the case 4.1c for 1 pu input voltage, V_{in} and n=1

4) Mode d: On the secondary side, as shown in mode d of Fig. 4.8, in the left leg, second step of FSC occurs: the conducting switch (S_{1a2}) of the incoming switch pair (S_{1a}) is gated ON (q_{1a2}) . In the left leg, even though S_{1a2} is gated ON, as the link voltage is positive, the diode D_{1a1} is reverse biased. Hence, the current continues to flow through the S_{3a1} and D_{3a2} pair, as D_{3a2} is forward biased. This mode of operation is different as compared to mode *iv* of Fig. 4.1b where the transition in the left leg was natural, unlike the forced commutation in this case. In case of the right leg switch pair, the third step of FSC occurs: the conducting switch (S_{2a1}) of the outgoing switch pair (S_{2a}) is gated OFF (q_{2a1}) and the current continues to flow through S_{4a2} and D_{4a1} pair as in mode c. Switches as shown in mode d of Fig. 4.8 conduct, resulting in an additional zero voltage, v_a interval as shown in mode d of Fig. 4.1c, different from mode *iv* of Fig. 4.1b. Fig. 4.6 shows the output voltage across Modules I, II and III followed by, v_a with switch non-idealities. The output voltage profile is similar to Fig. 4.5, except that it has an additional zero voltage interval.

The top plot of Fig. 4.9a, is the output voltage considering both the four-step commutation of the bi-directional switches and the dead-time of the primary side switches. This results in a large input current ripple as shown in the middle plot of Fig. 4.9a of one phase and bottom plot of Fig. 4.9a of all three phases. The high input ripple current will require large passive components on the input side. However, as explained earlier, by incorporating glitch compensation-I the output voltage improves as shown in top plot of Fig. 4.9b, which results in the reduction of the input current ripple as shown in the middle and the bottom plots of Fig. 4.9b of the single and three phases respectively. Hence, the modifications in the switching signals allow to improve the power quality.



Figure 4.9: (Top) Output phase voltage, v_a (a) without any glitch compensation, (b) with glitch compensation-I; (Middle) Input current of single H-bridge (a) without any glitch compensation, (b) with glitch compensation-I; (Bottom) Total input current of all three H-bridges (a) without any glitch compensation, (b) with glitch compensation-I



Figure 4.10: Switching signals of left leg bi-directional switches of all three AC-AC converter modules followed by the module and the phase-neutral output voltage: (a) ideal signals without four-step commutation; (b) with four-step commutation of secondary side bi-directional switches; (c) with four-step commutation of secondary side bi-directional switches and glitch compensation-II

As shown in Fig. 4.3 and 4.5 the glitches (highlighted by blue patches) in v_a are caused due to the interdependence of the three AC-AC converter modules in each phase. The section explains the cause and the compensation done to avoid the glitch. In Fig. 2.4 at a specific instant of time, consider the link voltages v_{sa1} , v_{sa2} and v_{sa3} and the output current, i_a to be positive. The output voltage level to be synthesized is $4nV_{in} \rightarrow 3nV_{in}$. As shown in Table 8.1, all the modules have to generate a pulsating voltage and hence, all three modules will undergo switching even though the link voltage is constant, as explained in the ideal modulation section. Fig. 4.10a shows the ideal switching signals of the bi-directional switches of the left leg of all the three modules in phase a as throughout the transition, in the right leg, the bottom bi-directional switch will be ON in all three modules. Also, the figure shows the output voltage waveform of the series connected converter modules followed by, v_a for the above described case. As shown in the Fig. 4.10a, when $q_{1a} = 1$; $q_{5a} = q_{9a} = 0$, v_a is $4nV_{in}$ and when $q_{1a} = 0$; $q_{2a} = q_{3a} = 1$, v_a is $3nV_{in}$. In case of ideal modulation, there is no delay during the switch transition and hence, there is no glitch in the output voltage.



Figure 4.11: Circuit diagram for the various modes of operation (left) after first step of FSC resulting in $4nV_{in}$ (center) after second step of FSC resulting in $7nV_{in}$ output voltage instead of $3nV_{in}$ (right) after second step of FSC with compensation resulting in $3nV_{in}$

Fig. 4.10b shows the switching signals for non-ideal bi-directional switches; that is, with FSC. In Module I, the transition is from the top two uni-directional switch set to the bottom, whereas for Module II and III the transition is vice-versa. Fig 4.11a shows the circuit diagram for all three modules when the output voltage, v_a is $4nV_{in}$ and the first step (turn OFF non-conducting switch of the outgoing switch pair) of FSC has already occurred. To illustrate, the non-conducting switches $(S_{1a2}, S_{7a1} \text{ and } S_{11a1})$ of the outgoing switch pair $(S_{1a}, S_{7a} \text{ and } S_{11a})$ respectively are gated OFF $(q_{1a2}, q_{7a1} \text{ and } q_{11a1})$. Fig. 4.11b shows the circuit diagram wherein, the conducting switches $(S_{3a2}, S_{5a1} \text{ and } S_{9a1})$ of the incoming switch pair $(S_{3a}, S_{5a} \text{ and } S_{9a})$ are gated ON $(q_{3a2}, q_{5a1} \text{ and } q_{9a1})$ simultaneously as shown in Fig. 4.10b. As the link voltage is positive, diode D_{3a1} is reverse biased, but $(D_{5a2} \text{ and } D_{9a2})$ are forward biased. Hence, in Module I no transition occurs. However, in Modules II and III the switching transition takes place. Hence, the output voltage is $7nV_{in}$ instead of $3nV_{in}$ as shown in Fig. 4.10b. When q_{1a1} is gated OFF, the output voltage v_{a1} across Module I is zero and the total output voltage v_a , is $3nV_{in}$. To summarize, Module I undergoes forced commutation whereas, Modules II and III undergo natural commutation resulting in a glitch in the output voltage.

The glitch in the output voltage can be avoided by delaying the signals of the modules undergoing natural commutation to match them with the signal of the module undergoing forced commutation. In this case, as shown in Fig. 4.11c, the conducting switch (S_{1a1}) of the outgoing switch pair (S_{1a}) is gated OFF (q_{1a1}) and the conducting switches $(S_{5a1} \text{ and } S_{9a1})$ of the incoming switch pair $(S_{5a} \text{ and } S_{9a})$ are gated ON $(q_{5a1} \text{ and } q_{9a1})$ simultaneously as shown in Fig. 4.10c. This causes Modules II and III to generate $2nV_{in}$ and nV_{in} respectively only when, v_{a1} is zero. Hence, the output voltage is $3nV_{in}$. However, as shown in Fig. 4.10c, the third and fourth step of FSC of modules undergoing natural commutation need to be further delayed.

From Table 8.1 and Fig. 4.5 it can be seen that, glitch in the phase-neutral output voltage occurs, whenever two or more secondary side modules change state simultaneously and the primary side remains fixed. Table 4.1, shows the switch which undergoes natural commutation (this switching is to be delayed to match with the switch undergoing forced commutation) for a particular output voltage level to be synthesized for a positive link voltage. Fig. 4.7 shows the output voltage across Modules I, II and III followed by the total phase-neutral output voltage, v_a with switch non-idealities after both compensation. The output voltage still goes to zero in each switching cycle caused because of the dead-time of primary side switches. The zero interval cannot be avoided but can be minimized by using devices with extremely fast turn on and turn off times.

Case	Output Voltage	Switch OFF (Forced)	Switch ON(Natural)
Ι	$6nV_{in} \rightarrow 5nV_{in}$ $2nV_{in} \rightarrow nV_{in}$	S_{5a1}	S_{9a1}
II	$5nV_{in} \to 6nV_{in}$ $nV_{in} \to 2nV_{in}$	S_{9a1}	S_{5a1}
III	$4nV_{in} \rightarrow 3nV_{in}$	S_{1a1}	S_{5a1}, S_{9a1}
IV	$3nV_{in} \rightarrow 4nV_{in}$	S_{5a1}, S_{9a1}	S_{1a1}

Table 4.1: Various instances of Glitch Compensation-II for a constant positive link voltage, for generating different positive levels in phase-neutral output voltage

4.2 Modulation-II: Incorporation of switch non-idealities

The previous chapter described the ideal modified hybrid modulation technique wherein, all the switches (that is no turn-on and turn-off times) are considered to be ideal and the output voltage profile is shown in Fig. 4.2 for a fundamental cycle. This section explains the distortions in the phase-to-neutral output voltage because of the modifications in the switching signals:

- i. considering only the switches of primary side H-bridge non-ideal.
- ii. to incorporate the four-step commutation strategy.
- iii. both (i) and (ii).

4.2.1 Dead-time in the switches of primary side H-bridge

As shown in Fig. 4.12b, a dead-time is incorporated in the primary side H-bridge switch signals. The switches of the AC-AC converter module are still considered ideal. For the case considered in the ideal modulation section, as shown in Fig. 4.12b, for the reason explained earlier, signals s_{1a} , s_{5a} and s_{10a} follow s_1 (same as p_1) and hence the switching signals are similar to the ideal modulation as shown in Fig. 4.12a. However, the switching signal p_2 (in this case s_2) is high after a dead-time interval. This implies, that the signals s_{2a} , s_{6a} and s_{11a} are also delayed as compared to the ideal modulation in Fig. 4.12a as these signals follow s_2 . But this will require the two bi-directional switches in each leg to switch simultaneously (S_{1a} turns OFF and S_{3a} turns ON). Due to the switch non-ideality a four-step commutation strategy is required.

4.2.2 Four-step commutation for switches of AC-AC Module

Each leg of the AC-AC converter module has to undergo a commutation process to satisfy two conditions:

- The isolated secondary winding connected to the AC-AC converter module is not short-circuited.
- The output current is not interrupted.

Implementation of the conventional four-step commutation strategy satisfies both the above conditions. As an example, when the link voltage is negative, if Module III is required to pulsate from $0 \rightarrow -nV_{dc}$, as explained in ideal modulation above, the output current, I_a which is assumed as shown in Fig. 4.13, has to switch from S_{11a} to S_{9a} . At a particular instant of time, the link voltage is negative and uni-directional switch pair, S_{11a1} and S_{11a2} are ON. A four-step commutation procedure is to be carried out to switch the currents from bi-directional switch S_{11a} to switch S_{9a} . The steps to do the same are described below:

In the first step, turn OFF S_{11a1} (outgoing non-conducting switch), as (S_{11a2}, D_{11a1}) pair is carrying the current.

Turn ON the IGBT, S_{9a1} (incoming conducting switch) after a commutation delay interval. As the link voltage is negative, diode D_{9a2} is reverse biased. Hence, the current will not commutate when S_{9a1} is turned ON but will continue to flow through (S_{11a2}, D_{11a1}) pair. Mode 4 of Fig. 4.13b shows the circuit diagram where D_{9a2} is reverse biased.

Turn OFF the IGBT, S_{11a2} (outgoing conducting switch), which was carrying the current after an additional commutation delay interval. At this instant, the current will commutate to (S_{9a1}, D_{9a2}) pair. This is forced commutation.

Turn ON the IGBT switch, S_{9a2} (incoming non-conducting switch) in which the diode D_{9a2} is carrying the current after one more commutation delay interval. In this manner, t]



Figure 4.12: (a) Ideal signals without dead-time of primary side H-bridge switches and without four-step commutation of secondary side bi-directional switches and ideal output voltage, v_a .(b) Signals with dead-time of primary side H-bridge switches and without four-step commutation of secondary side bi-directional switches. (c) Signals with dead-time of primary side H-bridge switches and with four-step commutation of secondary side bi-directional switches. (c) Signals with dead-time of primary side H-bridge switches and with four-step commutation of secondary side bi-directional switches.

In the above case the link voltage was constant and only the left leg of Module III went through the transition. But if the link voltage is also changing then the transition will occur in six modes of operation as depicted in Fig. 4.13.





 S_{1a1} mm D_{3a1} S_{4a1} S_{3a1} S_{3a2} Ą ۸ S_{5a1} See k ·0000 V_{dc} aaaa \diamondsuit D_{7a1} S7a1 S_{7a2} D_{7a2} ł $\Delta^{D_{9a1}}$ S_{9a1} S_{10a1} S_{9a2} 12 WWW S_{11a1} S_{12a1} D_{12a1} S_{11a2} S_{12a2} 3



(b)

 I_a



Figure 4.13: Circuit diagram for the various modes of operation for the case Fig. 4.12c. Blue indicates the gate signals and red indicates the current path.

The modes are explained by considering a case in which the desired output voltage to be generated is positive, $5nV_{dc}$ and the current direction is as shown in Fig. 4.13, and assumed to be positive while the link voltage pulsates from $+V_{dc}$ to $-V_{dc}$. Fig. 4.12c shows the switching signals of the primary side H-bridge switches $(s_1^* \text{ and } s_2^*)$ which are delayed by one commutation time to improve the output voltage profile and the two bi-directional switches of left leg of Module I (S_{1a} and S_{3a}) and Module III (S_{10a} and S_{11a}) followed by the phase-to-neutral output voltage for the above transition. As the bi-directional switch is not considered ideal, hence each uni-directional switch is controlled by one switching signal. As shown in Fig. 4.12c, n_{1a1} and n_{1a2} are the signals for uni-directional switch pair S_{1a1} and S_{1a2} respectively. The switching signals of the right leg bi-directional switches are not shown and can be interpreted from Fig.4.13 (Blue color indicates the switch turned ON and red indicates the current path). For the transition considered, as shown in Table 3.2, Module I and Module II will behave exactly the same and hence, the switching signals of Module II are not shown in Fig. 4.12c. A detailed description of the different modes of circuit operation is explained below:

Mode 1: This mode corresponds to the initial state as described above, wherein the link voltage is $+V_{dc}$ and the phase-to-neutral output voltage to be generated is $5nV_{dc}$ as

shown in Mode 1 of Fig. 4.12c. Mode 1 of Fig. 4.13a shows the circuit representation, which depicts the switches ON/OFF in each of the converter module to generate the above output voltage.

Mode 2: In this mode, s_1 goes low but as the actual primary side H-bridge signals are delayed by one commutation time, hence, the link voltage is still $+V_{dc}$. As mentioned earlier in ideal modulation, as the module output voltage to be generated in Modules I and II are positive and in Module III is negative, in Modules I and II the left leg top bi-directional switch signals and in Module III the right leg top bi-directional switch signal follow s_1 as shown in Fig. 4.12c. As s_1 goes low, the first step of the four-step commutation process (turn OFF the outgoing non-conducting switch) starts in left leg of Module I and II and in right leg of Module III. Whereas in the adjacent leg of each module there is no change in the switch state. As shown in Mode 2 of Fig. 4.13a, the current path in each of the three modules show that the phase-to-neutral output voltage will be maintained as in Mode 1 at $5nV_{dc}$.

Mode 3: As shown in Fig. 4.12c, on the primary side s_1^* goes low. This means none of the switches on the primary side H-bridge are ON. On the secondary side, the switches of the left leg of Modules I and II and right leg of Module III undergo second step of four-step commutation (turn ON the incoming conducting switch). At this instant, s_2 goes high and hence, the adjacent leg in each module undergo first step of the commutation process. As shown in Mode 3 of Fig. 4.12c, the phase-to-neutral output voltage will be zero instead of being maintained as in Mode 1. This is because as seen in Mode 3 of Fig. 4.13b, the switches of the left leg of Modules I and II and right leg of Module III undergo natural commutation. Also, in the adjacent leg of all the three modules, the bottom switch and the anti-parallel diode pair are still conducting. Hence, all the three modules freewheel the output current and hence, the output voltage is zero. Moreover, in the primary side H-bridge none of the switches/diodes conduct, and as the link current is zero, even the link voltage is zero.

Mode 4: At the beginning of the mode, as shown in Fig. 4.12c, on the primary side H-bridge s_2^* goes high. On the secondary side, as shown in Fig. 4.12c, left leg of Modules I and II and right leg of Module III undergo third step of four-step commutation (turn OFF the outgoing conducting switch) whereas, the adjacent leg in each module undergo second step of the commutation process. The phase-to-neutral output voltage will be positive but instead of being at $5nV_{dc}$, it will be at $6nV_{dc}$. This is because, the left leg bi-directional switches of Module I and II and the right leg bi-directional switch of Module III have already undergone commutation. In the right leg of Module I and II the bi-directional switches undergo natural commutation and hence, the module output voltage of Modules I and II are $4nV_{dc}$ and $2nV_{dc}$ respectively. But in Module III, as the link voltage is negative, the diode of the incoming bi-directional switch pair, D_{9a2} is reverse biased as explained in the four-step commutation section above, and Module III continues to generate zero voltage, whereas, the required voltage to be generated by Module III is $-nV_{dc}$.

Mode 5: In this mode, as shown in Fig. 4.12c, the last step of four-step commutation occurs in the left leg of Modules I and II and right leg of Module III (turn ON the incoming non-conducting switch), whereas, in the adjacent leg, third step of four-step commutation takes place. This is shown in Mode 5 of Fig. 4.13c. The phase-to-neutral output voltage will be $5nV_{dc}$ as shown in Fig. 4.12c, as all the modules generate the voltage as shown in Table 3.2 with the link voltage negative.

Mode 6: This mode corresponds to the final step of the switching transition. In this mode in the right leg of Modules I and II and in the left leg of Module III the last step of four-step commutation occurs. Mode 6 of Fig. 4.12c shows the switching signals that are high/low; and Mode 6 of Fig. 4.13c shows the circuit representation, which depicts the switches ON/OFF in each of the converter module so that the phase-to-neutral output voltage to be generated is $5nV_{dc}$ with the link voltage, $-V_{dc}$.

Hence, as shown in Fig. 4.12c the output voltage, v_a in comparison to the ideal case Fig. 4.12a, is distorted because of the switch non-idealities as explained in the six modes of operation.

Chapter 5

Design of High-Frequency Four Winding Transformer

5.1 Transformer design

For the converter topology shown in Fig. 2.4 the HFT requires a single primary winding and three secondary windings for each phase. The three secondary windings of the transformer should have an asymmetrical turns ratio as explained in the previous section along with low leakage inductance and ac resistance. Multi-winding HFTs using advanced soft magnetic materials, like amorphous and nanocrystalline have been designed for converter topologies to interface renewables to three-phase grid [85,86]. Although difficult to manufacture, several prototypes of multi-winding HFTs [83,85–88] and in [87] a 35kVA, 10kHz, 22kV/800V five-winding transformer has been designed and tested.

Table 5.1: Transformer specifications

Parameter	Real Power, P	Primary Voltage, V_p	Frequency, f	B_{max}	J	k_w
Value	$2 \ kW$	100 V	$15 \ kHz$	0.38 T	$3.5A/mm^2$	0.35

5.1.1 Transformer specifications and core material

Table 5.1 outlines the high-frequency transformer specifications. Area product method is used to design the transformer. The maximum flux density, B_{max} and current density J, are chosen to be 0.38 T and 3.5 A/mm^2 respectively [89]. Area product, A_p can be computed using (5.1) as all the other parameters like (V_{in} , I_{in} , f_{sw}) are known from the converter specifications. As shown in Fig. 5.2b, the volumes of the transformer core for different materials and high frequency have been compared. The soft magnetic materials allow to achieve lower volume for the same switching frequency. However, the transformer for the proposed converter prototype is designed with ferrite core 0P47228EC owing to the ease of availability. As the core dimensions are known, the core cross-sectional area can be computed. In the proposed converter topology the link voltages are square in nature. Hence, the required number of primary turns N_p , on a core with cross-sectional area A_c , and excited with a voltage, V_{in} at a link frequency f_{sw} so that the maximum flux density B_{max} is not exceeded can be determined.

5.1.2 Winding design

The winding design of a high-frequency transformer is a critical part of the design as the proper selection of windings ensure lower losses due to low ac resistance [90–93]. Also, proper winding arrangement will be essential to attain low leakage inductance. The skindepth at $f_{sw} = 15kHz$ is 0.532 mm. The primary and secondary windings were designed with foil conductors having dimensions 0.5588 mm X 6.35 mm and 0.2286 mm X 3.048 mm respectively. The bare copper wires were wound with 3M-44 tapes having an insulation thickness of 0.139 mm. The converter topology requires that the transformer should be designed such that the leakage inductances of the secondary side windings are low. For this, the 12 primary and 12, 24 and 48 secondary turns respectively of the four winding transformer are interleaved as shown in the winding arrangement diagram in Fig. 5.1b. The interleaving consists of four sections of primary and secondary turns.

$$A_p = \frac{2V_{in}I_{in}}{4f_{sw}B_{max}Jk_w} \tag{5.1}$$

Once, N_p is computed, the number of turns in the three secondary windings are $N_{s1} = 4nN_p$; $N_{s2} = 2nN_p$ and $N_{s3} = nN_p$ respectively; where n = 1 is considered for this design.



Figure 5.1: (a) 3-D model of the four winding transformer; (b) Interleaved winding technique of the transformer for low leakage inductance



Figure 5.2: (a) Prototype of the four winding transformer (b) HFT volume variation with core material and switching frequency

5.1.3 Inductance computation

Considering a pi-model of a four-winding transformer as shown in Fig. 5.3, the voltages across the four windings can be written as shown in (5.2), (5.3), (5.4) and (5.5) respectively.

$$v_1 = (L_{pa} + L_m)\dot{i_1} + L_m(n_{s1}\dot{i_2}) + L_m(n_{s2}\dot{i_3}) + L_m(n_{s3}\dot{i_4})$$
(5.2)

$$v_{2} = L_{m} n_{s1}^{2} \left(\frac{\dot{i}_{1}}{n_{s1}} \right) + (L_{la1} + L_{m} n_{s1}^{2}) \dot{i}_{2} + L_{m} n_{s1}^{2} \left(\frac{n_{s2}}{n_{s1}} \dot{i}_{3} \right) + L_{m} n_{s1}^{2} \left(\frac{n_{s3}}{n_{s1}} \dot{i}_{4} \right)$$
(5.3)

$$\begin{aligned}
\psi_3 &= L_m n_{s2}^2 \left(\frac{\dot{i_1}}{n_{s2}} \right) + L_m n_{s2}^2 \left(\frac{n_{s1}}{n_{s2}} \dot{i_2} \right) \\
&+ (L_{la2} + L_m n_{s2}^2) \dot{i_3} + L_m n_{s2}^2 \left(\frac{n_{s3}}{n_{s2}} \dot{i_4} \right) \end{aligned} (5.4)$$

$$v_{4} = L_{m} n_{s3}^{2} \left(\frac{\dot{i_{1}}}{n_{s3}} \right) + L_{m} n_{s3}^{2} \left(\frac{n_{s1}}{n_{s3}} \dot{i_{2}} \right) + L_{m} n_{s3}^{2} \left(\frac{n_{s2}}{n_{s3}} \dot{i_{3}} \right) + (L_{la3} + L_{m} n_{s3}^{2}) \dot{i_{4}}$$
(5.5)

$$\begin{bmatrix} v_1 \\ v_2 \\ v_3 \\ v_4 \end{bmatrix} = \begin{bmatrix} L_{pa} + L_m & n_{s1}L_m & n_{s2}L_m & n_{s3}L_m \\ n_{s1}L_m & L_{la1} + n_{s1}^2L_m & n_{s1}n_{s2}L_m & n_{s1}n_{s3}L_m \\ n_{s2}L_m & n_{s1}n_{s2}L_m & L_{la2} + n_{s2}^2L_m & n_{s2}n_{s3}L_m \\ n_{s3}L_m & n_{s1}n_{s3}L_m & n_{s2}n_{s3}L_m & L_{la3} + n_{s3}^2L_m \end{bmatrix} \begin{bmatrix} \dot{i_1} \\ \dot{i_2} \\ \dot{i_3} \\ \dot{i_4} \end{bmatrix}$$

ı

The leakage inductances of the transformer can be derived from the inductance matrix using the approximate pi-model of the transformer. Here, n_{sk} is $\frac{N_{sk}}{N_p}$, where, k is $\{1, 2, 3\}$. The designed transformer is characterized using AP Instruments Network Analyzer 102B. The bode plots with the secondary winding terminals open-circuited and short-circuited are shown in Fig. 5.4. The open circuit measurement is done across the terminals x_1 and x_2 with all the secondary winding terminals open. The short circuit measurement is done by shorting the terminals y_1 and y_2 . The inductance seen from x_1 and x_2 are $L_{pa} + L_{la1}/n_{s1}^2$. Six more measurements (shorting a terminal and recording the bode plot across another set of terminals) are done and from that the leakage inductance of all the four windings are computed. The leakage inductance values are: $L_{p1} = 0.312uH$; $L_{la1} = 0.41uH$; $L_{la2} = 0.662uH$; $L_{la3} = 1.086uH$ and the ac resistances are $R_{p1} = 22m\Omega$; $R_{la1} = 33m\Omega$; $R_{la2} = 112m\Omega$; $R_{la3} = 223m\Omega$.



Figure 5.3: Transformer pi-model considered for extracting the various transformer parameters [94]



Figure 5.4: Transformer characterization using network analyzer (a) open circuit measurement as seen across $x_1 x_2$ terminals with all secondary terminals open; (b) measurement across primary terminals $x_1 x_2$ with $y_1 y_2$ terminals short circuited in Fig. 5.3 and $y_3 y_4$, $y_5 y_6$ terminals open.

Chapter 6

System Level Analysis of Proposed Converter Topology

6.1 Closed loop control

In this chapter, the following system level analysis of the proposed converter topology will be demonstrated.

- Closed Loop Analysis in Simulation
- Operation under Module Failure
- Filter Inductor Design

Fig. 6.1 shows a simple closed-loop controller, implemented in simulation for the power flow control. As shown in Fig. 6.2, the input voltage and the power reference change at 0.1s and 0.2s respectively. The behavior of the converter to these changes is shown in the current magnitude which increases at 0.1s to maintain the same output power and reduce at 0.2s with the reduction in active power, P. Fig. 6.3 shows the dynamic response of the converter with the change in the input voltage at 0.1s. After a short transient, the system settles at the required operating point.





Figure 6.2: (Top) Three phase-neutral voltages and input voltage which changes at 0.1 sec, (Bottom) Three line currents along with output power command which changes at 0.2 sec



Figure 6.3: Zoomed in phase-neutral voltages and line currents at 0.1 sec which shows the dynamic behavior of the converter.

6.2 Operation under module failure

In case of a symmetrical multilevel converter, various fault-tolerant techniques have been investigated in the literature [95]. If any of the modules are faulted then a fundamental phase-shift compensation technique is used to restore balanced line currents for all three phases with reduced magnitude [96]. A set of non-linear equations as given in (6.1) are solved to generate the compensated angles between all three phases for balanced line currents. The Table in [96] outlines the angles for a eleven level inverter with five symmetrical modules in each phase. The fundamental phase-shift compensation (FPSC) technique is often used because of the easy implementation. However, the technique has some disadvantages as shown in [96]. For an asymmetrical converter, [97] shows a reconfiguration technique by using additional bi-directional switches in case of faults.

$$V_a^2 + V_b^2 - 2V_a V_b \cos\alpha = V_b^2 + V_c^2 - 2V_b V_c \cos\beta = V_c^2 + V_a^2 - 2V_c V_a \cos\gamma$$
(6.1)

$$\alpha + \beta + \gamma = 2\pi \tag{6.2}$$

The above FPSC technique has been investigated on the proposed topology. The angles can be computed using (6.1) for all cases. Fig. 6.4, 6.5 and 6.6 respectively show the phase, line voltages and the line currents for an unfaulted case.



Figure 6.4: Three phase voltages for unfaulted case



Figure 6.5: Three line-line voltages for unfaulted case



Figure 6.6: Three line currents for unfaulted case



and all modules healthy in the remaining phases. It can be seen from Fig. 6.7 that although v_c has only three levels due to only one module being healthy, using (6.1), as the angles are different from $\frac{2\pi}{3}$, the line voltages and currents as shown in Fig. 6.8 and 6.9 respectively are balanced with reduced magnitude.

Phase- a	Phase- b	Phase- c	Possible Solution (in degrees)
4, 2, 1	4, 2, 1	4, X, X	\checkmark {7,7,4} (α = 93.5; β = 133; γ = 133.5)
4, 2, 1	4, 2, 1	₩, 2, 1	\checkmark {7, 7, 3} ($\alpha = 85; \beta = 137; \gamma = 138$)
4, 2, 1	4, 2, 1	¥, ₹, 1	\checkmark {7,7,1} (α = 68.5; β = 145; γ = 146.5)
₩, 2, 1	₩, 2, 1	¥, ¥, 1	\checkmark {3,3,1} (α = 79.5; β = 140; γ = 140.5)

Table 6.1: Solution with reduced balanced currents for some fault cases



Figure 6.7: Modules $4nV_{in}$ and $2nV_{in}$ faulted in phase- c and all modules healthy in the remaining phases: Three phase voltages



Figure 6.8: Modules $4nV_{in}$ and $2nV_{in}$ faulted in phase- c and all modules healthy in the remaining phases: Three line-line voltages



Figure 6.9: Modules $4nV_{in}$ and $2nV_{in}$ faulted in phase- c and all modules healthy in the remaining phases: Three line currents

It is to be noted that, for a fault in $2nV_{in}$ and nV_{in} modules in phase-*c* and all modules healthy in remaining two phases, the angles computed using (6.1), will restore balanced currents with a higher per unit value in comparison to the above case even though phase-*c* has three voltage levels similar to Fig. 6.7. This is because, on this occasion, module $4nV_{in}$ is healthy and hence, the average voltage generated is higher. The per unit value of line current increases as shown in Fig. 6.12 in comparison to Fig. 6.9.



Figure 6.10: Modules $2nV_{in}$ and nV_{in} faulted in phase- c and all modules healthy in the remaining phases: Three phase voltages



Figure 6.11: Modules $2nV_{in}$ and nV_{in} faulted in phase- c and all modules healthy in the remaining phases: Three line-line voltages



Figure 6.12: Modules $2nV_{in}$ and nV_{in} faulted in phase- c and all modules healthy in the remaining phases: Three line currents

Table 6.1, provides a manner to restore balanced currents in all three phases in case of faulted modules for some cases. For the other cases, by solving (6.1) the appropriate angles can be computed. 4, 2, 1 indicates the three asymmetrical modules in phase- a. {7, 7, 3} indicates the number of positive levels that can be achieved. The tick mark indicates that the fundamental phase shift compensation technique applied to symmetrical configuration is valid and the corresponding angles are computed using (6.1) and provided in the Table 6.1. For other cases, not shown in Table 6.1, but shown in [96], healthy modules of certain phases are either connected to a lower healthy module or are disconnected to restore balanced currents.

6.3 Filter inductor design

6.3.1 Design procedure

Consider a three phase system with power P, the phase-to-neutral output voltage is v_a , and the switching frequency is f_s . The ripple current is 5% of the load current, I_a . The ripple current is inversely proportional to the filter inductance, L and the switching frequency, f_s . For the same ripple current, increasing the switching frequency will reduce the filter inductance, L.

In case of an ungapped core, the relative permeability of the material will reduce with increase in the output current, I_a and hence, the design of the inductor should be such that at the rated output current, I_a , the inductance is equal to L. To design a filter inductor with inductance, L the following procedure is followed:

• From a set of available cores (toroidal, U-core, EE-core) having different relative permeability, μ_r , determine the cross-sectional area, A_c and the magnetic path length,
l_m of the core.

• Compute the number of turns, N using the relation,

$$N = \sqrt{\frac{L_{start}l_m}{\mu_0\mu_r A_c}} \tag{6.3}$$

where, L_{start} is an initial inductance value chosen.

• Compute the magnetomotive force, H_{pk} at the maximum load current, I_a using,

$$H_{pk} = \frac{N\hat{I}_a}{l_m} \tag{6.4}$$

where, $\hat{I}_a = \sqrt{2}I_a$.

- Compute the relative permeability, $\mu_{r_{pk}}$ at this value of H_{pk} . Depending on the core manufacturer, either the variation of $\mu_{r_{pk}}$ versus H_{pk} will have a curve fitting relation [98, 99] or the value should be estimated from the permeability versus DC bias curve itself [100].
- Estimate the inductance from the computed $\mu_{r_{nk}}$, using the relation below:

$$L_{comp} = \frac{N^2 \mu_0 \mu_{r_{pk}} A_c}{l_m} \tag{6.5}$$

The computed inductance, L_{comp} will be the inductance at the rated current. If this inductance, $L_{comp} < L$, chose a larger core and increase L_{start} until $L_{comp}=L$.

6.3.2 Comparison of filter size for two modulation strategies

The converter has the dominant voltage harmonic at the sidebands of the switching frequency if modulated using modified uni-polar technique.

However, using the modified hybrid modulation technique shifts the dominant voltage harmonic to the sidebands of multiples of twice the switching frequency, hence the size of grid filter inductance reduces. As mentioned earlier, in case of this modulation technique, even though the switching frequency is f_s , the first dominant harmonic is at $2f_s$. Hence, for the same ripple current the frequency is double resulting in $L_{new} = L/2$. If the above procedure is followed, $L_{comp} = L/2$ which will result in a smaller core and hence a more compact grid filter inductor.

Chapter 7

Hardware Setup and Experimental Results

7.1 Hardware setup



Figure 7.1: Hardware prototype of three-phase isolated asymmetrical multilevel converter



Figure 7.2: Experimental Setup of proposed multilevel converter topology to Interface low voltage DC with three-phase grid

The proposed topology is simulated using MATLAB/Simulink and experimentally verified on a scaled down fifteen level three-phase laboratory prototype as shown in Fig. 7.1. Fig. 7.2 shows the hardware setup used to validate the proposed topology experimentally. Agilent 6575A is used as the input dc supply, V_{in} . The power modules from Microsemi (APTGL120TA120TPG) are used to design each of the three full-bridges for the threephases connected in parallel on the primary side. A total of 12 PWM signals are required for the three full-bridge modules. Three high-frequency transformers, one for each phase are designed as explained in the previous section. The power modules from Microsemi (APTGT50TDU60PG and APTGF50TDU120PG) are used to design the secondary side AC-AC converter module with bi-directional switches (two anti-parallel IGBTs connected in a common emitter configuration). Two power modules are used to design one AC-AC converter module. One module is for the two bi-directional switches in the top leg, (for example in Fig. 2.4, S_{1a} and S_{2a} in Module I are designed using one Microsemi module) and the other module is for the bottom leg switches $(S_{3a} \text{ and } S_{4a})$. The Module III with the lowest voltage (nV_{in}) in Fig. 7.1 is designed using two APTGT50TDU60PG power modules, whereas, Modules I and II $(4nV_{in} \text{ and } 2nV_{in} \text{ respectively})$ are designed using two APTGF50TDU120PG for each AC-AC converter module. All the three modules are connected in series on the output side. Similarly for phase, b and c, one full-bridge and three AC-AC converter modules are designed as shown in Fig. 7.1. 72 PWM signals are required for the secondary side switches forming the three-phase system. An FPGA-based control platform (Xilinx, XC3S500E) has been used to generate the PWM signals. For leakage commutation, a clamp circuit for each converter module was designed using four SiC schottky diodes (C4D05120A) in a full-bridge configuration having zero reverse recovery current and faster switching. Three LEM current sensors, one for each phase is used to aid in four-step commutation. Three LEM voltage sensors are used to sense the AC grid voltage so that the converter output voltages are generated at a phase shift angle, δ with respect to the source voltage.

7.2 Modulation-I: Simulation and experimental results

Tek PreVu Noise Filter Off 0 a T b 2 q1a1 3 q1a2 2 q2a1 q2a2 q3a1 **11** q3a2 . . . ₫ [q4a1 **[**]q4a2 İL ÎV : Ż Ш ٧ľ Í \square ٧a Z 8.00 V 100 V 400ns 31.9165kHz 12:02:23 m

7.2.1 Results of Glitch Compensation

Figure 7.3: Digital signals corresponding to the six modes of operation without any compensation followed by the phase-neutral output voltage, v_a in (CH-1) [100 V/div] in 400 ns/div time scale using the experimental setup similar to the simulation results as shown in Fig. 4.1a. Each commutation interval is 600 ns



Figure 7.4: Experimental results of the fundamental cycle of phase-neutral output voltage, v_a in (CH-1) [40 V/div] in 2 ms/div time scale similar to the simulation result as shown in Fig. 4.4



Figure 7.5: Digital signals corresponding to the six modes of operation with glitch compensation-I followed by the phase-neutral output voltage, v_a in (CH-4) [100 V/div] in 400 ns/div time scale using the experimental setup similar to the simulation results as shown in Fig. 4.1b



Figure 7.6: Experimental results of the fundamental cycle of phase-neutral output voltage, v_a in (CH-4) [40 V/div] in 2 ms/div time scale similar to the simulation result as shown in Fig. 4.5



Figure 7.7: Digital signals corresponding to the six modes of operation with glitch compensation-I followed by the phase-neutral output voltage, v_a in (CH-4) [100 V/div] and i_a in (CH-3) to show the reverse power flow in 400 ns/div time scale using the experimental setup similar to the simulation results as shown in Fig. 4.1c



Figure 7.8: Experimental results of the fundamental cycle of phase-neutral output voltage, v_a in (CH-2) [200 V/div], current, i_a (CH-4) [1 A/div], the source voltage, v_{ga} (CH-2) [200 V/div] in 4 ms/div time scale.



Figure 7.9: Digital signals corresponding to the left leg bi-directional switches of all three AC-AC converter modules followed by the phase-neutral output voltage, v_a in (CH-1) [40 V/div] in 1 μ s/div time scale using the experimental setup: without glitch compensation-II similar to the simulation results as shown in Fig. 4.10b



Figure 7.10: Digital signals corresponding to the left leg bi-directional switches of all three AC-AC converter modules followed by the phase-neutral output voltage, v_a in (CH-1) [40 V/div] in 1 μ s/div time scale using the experimental setup: with glitch compensation-II similar to the simulation results as shown in Fig. 4.10c

The first set of experimental results correspond to the output voltage profile with and without the various glitch compensation techniques as proposed in the modulation and commutation section. Fig. 7.3 shows the digital signals of the primary side switches and all the bi-directional switches of Module I corresponding to the six modes of operation as explained using Fig. 4.2 and Fig. 4.1a. As shown in Fig. 7.3, a voltage reversal is generated in mode II and zero voltage in mode III similar to the description in the non-ideal modulation section. Fig. 7.4 shows the output voltage, which is similar to the fundamental cycle of output voltage as shown in Fig. 4.4. Fig. 7.5 shows the digital signals of the primary side switches which are delayed by a commutation interval and all the bi-directional switches of Module I corresponding to the six modes of operation as explained using Fig. 4.3 and Fig. 4.1b. As shown in Fig. 7.5 the output voltage generates zero voltage in mode *iii* similar to the description in the glitch compensation- I section and does not generate a voltage reversal in mode *ii*. Fig. 7.6 shows the output voltage, which is similar to the fundamental cycle of output voltage as shown in Fig. 4.5. Fig. 7.7 shows the digital signals of the primary side switches which are delayed by a commutation interval and the bi-directional switches of left lef of Module I corresponding to the six modes of operation as explained using Fig. 4.8 and Fig. 4.1c. As shown in Fig. 7.7 the output voltage generates zero voltage in two modes similar to the description above rather than only in mode *iii* for the case of postive power flow from DC to three-phase grid. The reverse power flow, even though it has the same ideal signal generation due to the forced commutation generates a zero voltage for two commutation intervals as shown in Fig. 4.1c. However, the glitches

due to the interdependence of the three AC-AC converter modules are still not resolved (highlighted by the blue patches in Fig. 7.6) as explained in non-ideal modulation section. Fig. 7.9 shows the digital signals corresponding to the bi-directional switches in the left leg of all the three AC-AC converter modules in phase-a, followed by the output voltage, v_a . As shown in the Fig. 7.9, the output voltage, v_a has a glitch similar to Fig. 4.10b. With the glitch compensation- II implemented, as shown in Fig. 7.10, the rising edge of the signals, q_{5a1} and q_{9a1} are delayed to match with the falling edge of the signal, q_{1a1} . Also, the corresponding switch signals, q_{7a2} and q_{11a2} of Modules II and III are also delayed. This modification results in the mitigation of the glitch in the output voltage, v_a , similar to the simulation results as shown in Fig. 4.10c.

200 $v_{sa1}[V]$ 0 -200 200 $v_{sa2}[V]$ 0 -200 100 $v_{sa3}[V]$ 0 -100 50 $v_{pa}[V]$ 0 -50 0.0117 0.0118 time [s]

7.2.2 Results for a three-phase system

Figure 7.11: Simulated transformer link voltage across the three series connected modules, v_{sa1} in (CH-2) [200 V/div], v_{sa2} in (CH-3) [200 V/div] and v_{sa3} in (CH-4) [100 V/div] followed by the primary side link voltage, v_{pa} in (CH-1) [50 V/div] in 10 μ s/div time scale.



Figure 7.12: Experimental transformer link voltage across the three series connected modules, v_{sa1} in (CH-2) [200 V/div], v_{sa2} in (CH-3) [200 V/div] and v_{sa3} in (CH-4) [100 V/div] followed by the primary side link voltage, v_{pa} in (CH-1) [50 V/div] in 10 μ s/div time scale.



Figure 7.13: Simulated module output voltage of the three series connected modules, v_{a1} in (CH-1) [200 V/div], v_{a2} in (CH-2) [100 V/div] and v_{a3} in (CH-3) [1000 V/div] followed by the phase-neutral output voltage, v_a (CH-4) [200 V/div] in 2 ms/div time scale.



Figure 7.14: Experimental module output voltage of the three series connected modules, v_{a1} in (CH-1) [200 V/div], v_{a2} in (CH-2) [100 V/div] and v_{a3} in (CH-3) [1000 V/div] followed by the phase-neutral output voltage, v_a (CH-4) [200 V/div] in 2 ms/div time scale.



Figure 7.15: Experimental line current (CH-1) [1 A/div] along with the phase-neutral output voltage of all three-phases, v_a (CH-2), v_b (CH-3) and v_c (CH-4) [100 V/div] in 4 ms/div time scale.



Figure 7.16: Experimental currents of all three-phases, i_a (CH-1), i_b (CH-3) and i_c (CH-4) [0.5 A/div] in 4 ms/div time scale.



Figure 7.17: Simulated transformer link voltage, v_{sa1} in (CH-1) [400 V/div] followed by the transformer link currents through the three secondary windings, i_{sa1} in (CH-2) [2 A/div], i_{sa2} in (CH-3) [2 A/div] and i_{sa3} in (CH-4) [2 A/div] in 2 ms/div time scale.



Figure 7.18: Experimental transformer link voltage, v_{sa1} in (CH-1) [400 V/div] followed by the transformer link currents through the three secondary windings, i_{sa1} in (CH-2) [2 A/div], i_{sa2} in (CH-3) [2 A/div] and i_{sa3} in (CH-4) [2 A/div] in 2 ms/div time scale.



Figure 7.19: Simulated transformer link voltage, v_{sa1} in (CH-1) [400 V/div] followed by the transformer link currents through the three secondary windings, i_{sa1} in (CH-2) [2 A/div], i_{sa2} in (CH-3) [2 A/div] and i_{sa3} in (CH-4) [2 A/div] in 40 μ s/div time scale.



Figure 7.20: Experimental transformer link voltage, v_{sa1} in (CH-1) [400 V/div] followed by the transformer link currents through the three secondary windings, i_{sa1} in (CH-2) [2 A/div], i_{sa2} in (CH-3) [2 A/div] and i_{sa3} in (CH-4) [2 A/div] in 40 μ s/div time scale.

The simulation and experimental results of the transformer link voltages across the primary and three secondary windings are shown in Fig. 7.11 and Fig. 7.12 respectively. The complimentary switching of the input side full-bridge at 15kHz with a dead-time of 600ns generates the primary side link voltage, v_{pa} . As shown in those figures, for an input voltage, V_{in} of 50V; the transformer secondary side link voltages $(v_{sa1}: v_{sa2}: v_{sa3})$ are 200V, 100V and 50V in a ratio of (4:2:1). Fig. 7.13 and Fig. 7.14 show the simulation and experimental results of the output voltages across each module, v_{sa1} , v_{sa2} and v_{sa3} followed by the total phase-neutral output voltage, v_a . The differences in the magnitude of the output voltage between simulation and experimental results, is primarily due to the device drop across each switch which is not considered in the simulation model. As shown in the figures, all the modules generate a zero voltage during the dead-time of primary side full-bridge. Also, the top module is used to generate the top four output voltage levels and freewheels the output current for the remaining three levels. Fig. 7.15 shows the line current, i_a followed by the phase-neutral output voltage of all three-phases, v_a , v_b and v_c . As explained earlier, in all three-phases, the output voltage goes to zero when the link voltage switches. Fig. 7.16 shows the line currents of the three-phase system.

Fig. 7.17 and Fig. 7.18 shows the link voltage and three link currents through the transformer secondary windings. The link current, i_{sa1} is zero at some intervals in a fundamental cycle. This is because as explained above, for generating the lower output voltage levels, Module I will freewheel the output current and hence, the link current, i_{sa1} through the transformer winding is zero. Similarly, the current through the other secondary winding will be zero whenever Module II freewheels the current. Figs. 7.19 and 7.20 shows the same above plots on a 40 μ s/div timescale. The link currents, i_{sa1} and i_{sa2} follow the link voltage, v_{sa1} , but, as Module I switches even though the link voltage is constant, the link current, i_{sa3} also switches.

7.2.3 Results for bi-directional power flow

$$P = 3 \frac{v_{a1} v_{ga}}{X_l} sin\delta \tag{7.1}$$

$$m_a = \frac{\hat{V}_{a,1}}{7nV_{in}} \tag{7.2}$$

The mathematical relation for the power transmission and the voltage conversion are given in (7.1) and (7.2) respectively. Here, $X_l=2\pi$ fsL and $\hat{V}_{a,1}$ is the peak value of the fundamental output voltage.

The line-line rms voltage of the supply is 130V, and v_a , the converter output voltage is generated at a specific angle of, $\delta \pm 0.25$ rad with respect to the source voltage, v_{ga} . The series inductor, L_a , of approximate value is used to limit the power flow. The resulting output current i_a , output voltage, v_a for the power flow in either direction are shown in Figs. 7.21 and 7.22. The harmonic spectrum of the multilevel output voltage, v_a is shown in Fig. 7.23. The high number of levels results in a THD of 18% and the dominant harmonics are at the multiples of switching frequency. This allows to achieve low output current THD of less than 5%, with the harmonic spectrum as shown in Fig. 7.24 and hence a high power quality.



Figure 7.21: Experimental result of the converter output voltage, v_a , (CH-3) [200 V/div], current, i_a (CH-2) [1 A/div], the source voltage, v_{ga} (CH-1) [250 V/div] and source current [5 A/div] in 4 ms/div time scale



Figure 7.22: Experimental result of the converter output voltage, v_a , (CH-3) [200 V/div], current, i_a (CH-2) [1 A/div], the source voltage, v_{ga} (CH-1) [250 V/div] and source current [5 A/div] in 4 ms/div time scale.



Figure 7.23: Harmonic spectrum of output phase voltage, v_a : Modulation-I



Figure 7.24: Harmonic spectrum of the output current, i_a with a THD of 2.5%.

7.2.4 Results for operation under module failure

A particular scenario is considered in which in phase-a and b, $4nV_{in}$ modules are faulted and in phase- c, nV_{in} module is faulted. Fig. 7.25 shows the output voltage and current for both the unfaulted phase- b and faulted phase- c respectively. The maximum modulation index that can be achieved in this case is 0.42. In phase- a and b, the output voltage is generated using modules II and III. In case of phase- c wherein, module-III is faulted, in order to generate balanced sinusoidal output currents modules-I and II are operated at a reduced modulation index of 0.2 to generate the same rms voltage resulting in balanced currents. The THD of the output voltage, v_c of the faulted phase- c is 48.9 % as shown in Fig. 7.26 which is much higher than that of phase- b output voltage of 26.6 %. Although the THD of output voltage of the faulted phase- c is high, the technique results in balanced output currents for both the phases as shown in Fig. 7.25. For an input dc voltage of 50V, the peak of the fundamental output voltage of the unfaulted phase is 110.2V, and 20.76V ripple rms voltage; whereas for the faulted phase it is 116.6V and 40.4V respectively. This results in a THD of 26.6% and 48.9% respectively of phase b and c. However, both the phases have a fundamental peak current of 0.55A.



Figure 7.25: Experimental result of the output voltage, v_b in (CH-1) [100 V/div] and output current, i_b in (CH-2) [0.5A/div] of phase-*b* with $4nV_{in}$ module faulted and of v_c in (CH-3) [100V/div], current, i_c in (CH-4) [0.5 A/div] of phase-*c* with n_{in} module faulted, in 4 ms/div time scale for a modulation index of 0.4



Figure 7.26: Harmonic spectrum of the output voltage, v_c of the faulted phase- c



Figure 7.27: Harmonic spectrum of the output current, i_c of the faulted phase- c

7.3 Modulation-II: Simulation and experimental results

The modulation technique is verified on a scaled down fifteen level three-phase laboratory prototype as shown in Fig. 7.1. This section presents a detailed description of the hardware prototype along with key experimental and simulation results.

Table 7.1: Specifications

V_{in}	I_a	I_{in}	$\widehat{V_{ll1}}$	f_{sw}	f_o	R_{load}	L_{load}
$45 \mathrm{V}$	0.96 A	15.4 A	471.1 V	$15 \mathrm{~kHz}$	60 Hz	$200~\Omega$	$46.6 \mathrm{mH}$

The proposed modulation technique on the converter topology was simulated using MATLAB/Simulink. The simulation parameters are same as the hardware prototype and the results are presented together. In the commutation section, it was mentioned that, due to the dead-time of the primary side H-bridge switches the output voltage will be zero whenever, the link voltage changes polarity. Fig. 7.29 provides the experimental validation of this concept described in the commutation section. The simulation and experimental results of the output current, i_a ; phase-to-neutral output voltage, v_a and transformer link voltage v_{sa1} , at the instants when v_{sa1} goes to zero are shown in Fig. 7.28 and Fig. 7.29 respectively. As shown in the figures, the output load current, i_a varying at 60Hz is constant on a 10 μ s timescale. The secondary side link voltage, v_{sa1} will not transition instantly from +172V to -172V because of the dead-time of the primary side H-bridge switches. As shown in Mode 3 of Fig. 4.13b in the commutation section, during the dead-time all the switches of

the primary side H-bridge are gated OFF. At this instant, the continuous inductive output current, i_a freewheels through the AC-AC converter modules resulting in the output voltage, v_a to be zero as shown in Fig. 7.29.

In the ideal modulation section, it was mentioned that the modules which generate a constant output voltage switch only when the link voltage switches. But, the module which has to generate a pulsating output voltage undergoes switching transition even though the link voltage is constant. The next result demonstrates the above description using the experimental setup. Figs. 7.30 and 7.31 show the output line current, i_a , Module I output voltage, v_{a1} , Module III output voltage, v_{a3} and Module I link voltage, $v_{sa1}=\pm 172$ V using simulation and experimental setup respectively. As explained in the ideal modulation section, in order to generate $7nV_{in} \leftrightarrow 6nV_{in}$, that is $315V \leftrightarrow 270V$ output voltage, Module I which generates a constant voltage of, $4nV_{in}$, that is 164V (after the device drops); switches only when the link voltage changes polarity. Ideally, v_{a1} should be constant at 164V as shown in Fig. 4.2, but due to switch non-idealities as explained using Fig. 7.29 and Mode 3 of Fig. 4.13b, v_{a1} will momentarily generate zero voltage during the dead-time of primary side H-bridge switches. Module III in order to generate a pulsating voltage $nV_{in} \leftrightarrow 0$, that is, $36V \leftrightarrow 0$, undergoes switching even though the link voltage, v_{sa1} is constant as shown in Fig. 7.31



Figure 7.28: Simulation results of line current, i_a ; Phase-to-neutral output voltage, v_a and transformer link voltage v_{sa1}



Figure 7.29: Experimental results of line current, i_a (CH-2), with 0.5 multiplication factor [5 A/div]; Phase-to-neutral output voltage, v_a (CH-3) [200 V/div] and Transformer link voltage v_{sa1} (CH-2) [200 V/div], in 10 μ s/div time scale



Figure 7.30: Simulation results of line current, i_a ; module output voltages, v_{a1} and v_{a3} in modules I and III respectively and transformer link voltage v_{sa1}



Figure 7.31: Experimental results of line current, i_a (CH-2), with 0.5 multiplication factor [5 A/div]; module output voltages, v_{a1} (CH-4) [200 V/div] and v_{a3} (CH-3) [40 V/div] in modules I and III respectively and transformer link voltage v_{sa1} (CH-2) [400 V/div] in 10 μ s/div time scale



Figure 7.32: Digital signals corresponding to the six modes of operation as shown in Fig. 4.12c followed by the phase-to-neutral output voltage.



Figure 7.33: Output voltage of the three series connected modules, v_{a1} in (CH-4) [50 V/div], v_{a2} in (CH-3) [40 V/div] and v_{a3} in (CH-2) [10 V/div] followed by the phase-to-neutral output voltage, v_a in (CH-1) [100 V/div] in 4 μ s/div time scale.

Fig. 7.32 shows the digital signals corresponding to the six modes of operation as described in commutation section using Fig. 4.12c. Fig. 7.33 shows the output voltage across each module followed by the total output voltage, v_a . As shown in Fig. 7.33, in order to generate v_a of $5nV_{in}$; $v_{a1}=4nV_{in}$, $v_{a2}=2nV_{in}$ and $v_{a3}=-nV_{in}$ even though the link voltage transitions from $+V_{in}$ to $-V_{in}$. The output voltage, v_a in Fig. 7.32 is delayed in comparison to the ten digital signals. As explained in the six modes of operation in the commutation section, the voltage, v_a will be $5nV_{in}$ in modes 1 and 2 and then it will generate a zero voltage in mode 3 for the aforementioned reason explained using Fig. 7.29. However, after a zero voltage, instead of generating $5nV_{in}$, output voltage is $6nV_{in}$ in mode 4. This is because, the switching of module output voltage, v_{a3} as compared to v_{a2} and v_{a1} is delayed as explained in mode 4 of Fig. 4.13b. The module output voltages v_{a1} and v_{a2} of Modules I and II respectively undergo natural commutation however, the module output voltage, v_{a3} will undergo forced commutation that is, switch from $0 \rightarrow -nV_{in}$ in the third step of the four-step commutation strategy. Hence, the resulting voltage v_a will be $6nV_{in}$ for one commutation step as shown in Fig. 7.32. Finally in mode 5 and 6 the output voltage, v_a is again $5nV_{in}$.

Figs. 7.34 and 7.35 respectively show the simulation and experimental results of the output voltage, v_a , output line current, i_a and two module output voltages, v_{a1} and v_{a2} when an R-L load is connected to the system. As shown in the figures, each module output voltage is zero during the time the link voltage changes polarity as shown in Fig. 7.29. For a particular output voltage level generation, switches in each module will undergo natural/forced commutation depending on the output current, i_a as explained using different modes of circuit operation in the commutation section and shown experimentally in Fig.

7.32. Hence, the output voltage, v_a in Fig. 7.35 has a profile different as compared to Fig. 4.2 generated using ideal switching signals. The simulation and experimental results along with the switch non-ideality also have the transformer leakage inductance. This will require a clamp circuit for leakage commutation. The modulation index considered was 0.97 and the input dc voltage was 45V. The link voltages of the three secondary windings after the device drops are 172V, 86V and 43V respectively as shown in Fig. 7.12. The module output voltages after the device drop of the AC-AC converter module switches are 164V, 80V and 36V respectively. This results in a peak output voltage of 280V. The experimental peak value of the fundamental output voltage, $\hat{V}_{a,1}$ was computed as 272V. This results in a load current of 0.96A as shown in Fig. 7.35. According to the mathematical relation in (7.2), $\hat{V}_{a,1}$ is 305.5V, which should have corresponded to a current of 1.07A. The reason for the mismatch of 0.11A between the analytical and the experimental result is due to the device voltage drop across the switches as explained above. Due to the switch non-ideality and the transformer leakage inductance the THD in the phase-to-neutral output voltage, v_a is 15.84 %.



Figure 7.34: Simulation results of phase-to-neutral output voltage, v_a along with the line current followed by v_{a1} and v_{a2}



Figure 7.35: Experimental results of phase-to-neutral output voltage, v_a (CH-1) [100 V/div] along with the line current (CH-4) [2 A/div] followed by v_{a1} in (CH-3) [200 V/div] and v_{a2} in (CH-2) [100 V/div] in 4 ms/div time scale.



Figure 7.36: Phase-to-neutral voltages, v_a (CH-3), v_b (CH-1) and v_c (CH-4) [200 V/div]; along with the line current (CH-2), with 0.5 multiplication factor [5 A/div] in 4 ms/div time scale.



Figure 7.37: Line-to-Line output voltage of all three phases, v_{ab} (CH-3), v_{bc} (CH-1) and v_{ca} (CH-4) [200 V/div]; along with the line current (CH-2), with 0.5 multiplication factor [5 A/div] in 4 ms/div time scale.

Fig. 7.36 shows the experimental results of the line current, i_a followed by the phaseto-neutral output voltage of all three phases, v_a , v_b and v_c . Fig. 7.37 shows the line-to-line output voltage of the three-phase system.

Fig. 7.38 shows the experimental results of the leg voltages, v_{l1} and v_{l2} of left and right leg respectively of Module I followed by the module output voltage, v_{a1} for an input voltage of 45V. As shown in the figure, both the leg voltages switch at the link frequency to generate the module output voltage, v_{a1} as shown in Table 3.2. The module output voltage, v_{a1} in Fig. 7.38 generates a zero voltage which is not the case in the ideal output voltage generated using ideal signals. However, the harmonic spectrum as shown in Fig. 7.39 is similar to the harmonic spectrum using ideal signals and the module output voltage has dominant harmonics at the sidebands of twice the switching frequency which is 30kHz. Fig. 7.43 shows the experimental output voltage, v_a generated using the proposed modulation. The output generates a zero voltage whenever the primary side H-bridge switches, due to the switch non-idealities. The harmonic spectrum of v_a in Fig. 7.45 shows that the dominant harmonics are shifted to the sidebands of multiples of twice the switching frequency. The results are consistent with the literature [69,71]. Fig. 7.39 shows the harmonic spectrum of the leg voltages and the module output voltage, v_{a1} . As shown in the figure, similar to the harmonic spectrum using ideal signals, both the leg voltages have dominant harmonics at the switching frequency of 15kHz and its sidebands. However, the differential voltage, v_{a1} has the dominant harmonic at the sidebands of twice the switching frequency which is 30kHz. Fig. 7.40 shows the nature of the leg voltages and the module output voltage, during the PWM operation; $4nV_{in} \leftrightarrow 0$. As shown in the figure, the module output voltage, v_{a1} because of the differential voltage across the two legs of Module I switches twice in a 15kHz interval. Fig. 7.41, shows the harmonic spectrum of these voltages at 15kHz and its sidebands. In the figure, the normalized amplitude at each harmonic frequency from 14.8kHz-15.2kHz is shown on the left axis(in blue) and the phase angle in radians (in red) is shown on the right axis. Both the leg voltages, v_{l1} and v_{l2} have almost the same normalized amplitude at 15kHz and the sidebands. However, as the module output voltage is a difference of the two leg voltages, the same is not reflected in the harmonic spectrum of v_{a1} as shown in Fig. 7.41.



Figure 7.38: Fundamental cycle of leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1}



Figure 7.39: Harmonic spectrum of leg voltages, v_{l1} , v_{l2} of Module I followed by the module output voltage, v_{a1}



Figure 7.40: Zoomed in module voltage to show the profile during PWM operation



Figure 7.41: Harmonic spectrum at 15kHz to show the cancellation of the switching frequency harmonic component in the module output voltage, v_{a1} .



Figure 7.42: Modulation I: Module output voltage of the three series connected modules, v_{a1} in (CH-1) [200 V/div], v_{a2} in (CH-2) [100 V/div] and v_{a3} in (CH-3) [1000 V/div] followed by the phase-neutral output voltage, v_a (CH-4) [200 V/div] in 2 ms/div time scale, with a switching frequency of 15 kHz.



Figure 7.43: Modulation I: Module output voltage of the three series connected modules, v_{a1} in (CH-1) [200 V/div], v_{a2} in (CH-2) [100 V/div] and v_{a3} in (CH-3) [1000 V/div] followed by the phase-neutral output voltage, v_a (CH-4) [200 V/div] in 2 ms/div time scale, with a switching frequency of 15 kHz.



Figure 7.44: Modulation I: Harmonic spectrum of the experimental phase-neutral output voltage, v_a with a switching frequency of 15 kHz.



Figure 7.45: Modulation II: Harmonic spectrum of the experimental phase-neutral output voltage, v_a with a switching frequency of 15 kHz.

Chapter 8

High-Frequency Link Asymmetrical Converter

8.1 Converter topology



Figure 8.1: Converter Topology to Interface low voltage DC with three-phase Grid.

8.1.1 Circuit description

The converter topology with high-frequency transformer isolation is shown in Fig. 8.1. The asymmetrical converters comprising of switches S_1 - S_{12} are connected in parallel on the input side in order to share the high input currents. I_{in} is the total input current from the parallel connection of all three converter modules. Each of the three asymmetrical converters are connected to three high-frequency transformers which have asymmetrical turns ratio of 1:4n,

1:2n and 1:n respectively. I_{p1} , I_{p2} and I_{p3} are the primary currents of the three HFTs. V_{s1} , V_{s2} and V_{s3} are the transformer voltages across each of the three secondary windings of the HFT and V_s is the summation of these voltages which is applied across the diode bridge. The converter modules are switched so as to generate a high-frequency stepped voltage across the output of each of the transformer secondary resulting in a multi-level stepped waveform across the diode bridge. Power flow of PV being unidirectional, secondary side of the HFT is connected to a diode bridge rectifier D_1 - D_4 and a CSI made of switches S_a - S'_c via a dc link inductor, L_{dc} . The diode bridge and PWM CSI interface a three phase medium voltage grid. I_{dc} is the dc link inductor current. The CSI is connected to the grid via a capacitor bank, which is required for the current commutation, because of the presence of the grid source inductance. I_c is the current through the capacitor and I_g is the grid current.

8.1.2 Modulation strategy of input side converter

The primary side asymmetrical converters are switched so as to generate multi-level stepped voltage across the transformer secondary winding. With only three modules, 15 levels in the transformer secondary voltage can be generated; that is 7 positive levels, 7 negative levels and one zero level.

Zone	Modulation Index	Module I	Module II	Module III
Zone I	1-0.8571	1	1	1
Zone II	0.8571-0.7143	1	1	0
Zone III	0.7143-0.5714	1	0	1
Zone IV	0.5714-0.4286	1	0	0
Zone V	0.4286-0.2857	0	1	1
Zone VI	0.2857-0.1428	0	1	0
Zone VII	0.1428-0	0	0	1

Table 8.1: Modulation Strategy for Different Positive Levels for V_s

Generating the output voltage in zone I: For zone I operation the modulation index is from 1-0.8571. The switches S_1 , S_4 , S_5 , S_8 , S_9 and S_{12} are switched ON throughout the zone I operation to generate a constant voltage denoted by '1' in Table 8.1. Table 8.1 shows the modulation strategy for all the 7 zones of operation. In Table 8.1, '0' refers to the state in which the switches are switched so as to generate 0 voltage. If for all of the 7 zones of operation as shown in Table 8.1, switching the complimentary switches will generate the 7 negative levels.

The pulses can be generated using 7 level shifted carrier signals and comparing them with a reference sinusoidal signal. The seven carrier pulses hence obtained are XOR-ed accordingly to generate the pulses for the three converter modules. Fig. 8.3a shows the output voltage across the three secondaries followed by the total voltage depicting the 15 levels.

8.1.3 HFT voltage rating analysis based on DC link current of CSI

Under a unity power factor operation at the three phase grid, the capacitor current has to be supplied by the CSI [101]. Hence, the minimum DC link current I_{dc} can be expressed as in (8.1). Also, the dc link voltage across the output of the diode bridge, V_{dc} can be expressed in terms of the fundamental peak value of the total output secondary terminal voltage, \hat{V}_{st}^1 as given in (8.2). Assuming the CSI to be made of ideal switches, $V_{dc}I_{dc}=P_g$. Hence, from this P_g can be written as in (8.3):

$$I_{dc} = \frac{\sqrt{\left(\frac{2P_g}{3\hat{V}_g}\right)^2 + (2\pi f_s C_i \hat{v}_c)^2}}{m_i} \tag{8.1}$$

$$V_{dc} = \frac{2}{\pi} \hat{V}_{st}^1 \tag{8.2}$$

$$P_g = \frac{2}{\pi} I_{dc} \hat{V}_{st}^1 \tag{8.3}$$

In case of an ideal transformer, \hat{V}_{st}^1 is the same as \hat{V}_s^1 and can be expressed as in (8.4):

$$\hat{V}_{st}^1 = \hat{V}_s^1 = \hat{V}_{s1}^1 + \hat{V}_{s2}^1 + \hat{V}_{s3}^1 \tag{8.4}$$

where, \hat{V}_{si}^1 , $i = \{1, 2, 3\}$ are the peak values of the fundamental component of each of the three transformer secondaries. Hence, (8.2) can be written as in (8.5)

$$\hat{V}_s^1 = \frac{\pi}{2} V_{dc} \tag{8.5}$$

As \hat{V}_s^1 is known, the transformer secondary voltages can be obtained from the following relation:

$$\hat{V}_{s1} = 0.57143 (= 4/7) \hat{V}_s^1$$

 $\hat{V}_{s2} = 0.28571 (= 2/7) \hat{V}_s^1$

$$\hat{V}_{s3} = 0.14286 (= 1/7) \hat{V}_s^1$$

where, $\hat{V}_{s1} = 2\hat{V}_{s2} = 4\hat{V}_{s3} = 4n\hat{V}_p$. The transformer turns ratio are n:2n:4n, and hence, the transformer secondary winding with 4n turns ratio will need to generate a voltage 4/7times \hat{V}_s^1 . As the transformer secondary voltages are known, from simulation the rms values: V_{s1}^{rms} , V_{s2}^{rms} and V_{s3}^{rms} are determined and can be used to determine the apparent power rating of the three high power high-frequency transformers using (8.6).

$$S_{link} = I_s^{rms} [V_{s1}^{rms} + V_{s2}^{rms} + V_{s3}^{rms}]$$
(8.6)

8.1.4 Effect of transformer leakage inductance

For a non-ideal transformer, where $L_{lkgtot} = \sum_{i=1}^{3} L_{si} + n_i^2 L_{pi}$ and $R_{tot} = \sum_{i=1}^{3} R_{si} + n_i^2 R_{pi}$, $n_i = \frac{N_{si}}{N_p}$ wherein, L_{pi} , L_{si} , R_{pi} , and R_{si} , $i = \{1, 2, 3\}$ are the primary and secondary leakage inductances and ac resistances respectively of the three HFTs:

$$\hat{V}_{st}^1 \neq \hat{V}_s^1 \tag{8.7}$$

and,

$$\hat{V}_{s}^{1} = \frac{\pi}{2} \left[V_{dc} + \frac{2}{\pi} I_{dc} (2\pi f_{link} L_{lkgtot} + R_{tot}) \right]$$
(8.8)

Hence, P_g can be written as in (8.9) and from this the fundamental peak value of the total transformer secondary voltage, \hat{V}_s^1 can be computed in terms of the dc link current, I_{dc} and grid power, P_g as given in (8.10).

$$P_g = I_{dc} \left[\frac{2}{\pi} \hat{V}_s^1 - \frac{2I_{dc}}{\pi} (2\pi f_{link} L_{lkgtot} + R_{tot}) \right]$$
(8.9)

$$\hat{V}_{s}^{1} = \frac{2I_{dc}^{2}(2\pi f_{link}L_{lkgtot} + R_{tot}) + \pi P_{g}}{2I_{dc}}$$
(8.10)

In case of a non-ideal transformer, according to (8.7), the terminal voltage is not the same as \hat{V}_s^1 . For the same \hat{V}_s^1 as in the case of an ideal transformer, the presence of transformer leakage inductance and ac winding resistance will result in a reduction of the terminal voltage, \hat{V}_{st}^1 . Hence, according to (8.2) the dc link voltage, V_{dc} and dc link power P_{dc} will also be affected.

Parameters	$L_{lkgtot} = 0$		L_{lkgtot}	$=50\mu H$	$L_{lkgtot} = 100 \mu H$	
-	Analytical	Simulation	Analytical	Simulation	Analytical	Simulation
\hat{V}_s^1	5814.2 V	5813.7 V	6238.6 V	6237.8 V	6663 V	6667.9 V
V_{dc}	3701.5 V	3697 V	3701.5 V	3700 V	3701.5 V	3620 V
I_s^{rms}	-	$130.15 \ A$	-	127.16 A	-	124.36 A
P_{dc}	$500 \ kW$	$499.1 \ kW$	$500 \ kW$	$498.5 \ kW$	$500 \ kW$	$489 \ kW$
S_{link}	-	$603.9 \ kVA$	-	$633.04 \ kVA$	-	$661.33 \ kVA$

Table 8.2: Transformer Power Rating variation with the Leakage Inductance

So to ensure that the dc link power, P_{dc} is same as in the case of an ideal transformer, the transformer voltage rating has to be increased. The increased total transformer secondary voltage can be computed using (8.10). With the value of \hat{V}_s^1 known, as explained in the previous section the rms voltages of individual transformer secondary windings can be determined.

The presence of the transformer leakage inductance, causes a phase shift between the fundamental current component and the fundamental voltage component, resulting in reactive losses. The resultant lagging power factor increases the overall transformer power rating. Table 8.2 summarizes the increase in the transformer power rating caused due to the presence of leakage inductance in comparison to an ideal transformer. Three different values of leakage inductance are chosen: 0 (assuming an ideal transformer) 50 μH and 100 μH . For all the cases, values of the parameters like \hat{V}_s^1 , V_{dc} , I_s^{rms} , P_{dc} and P_{link} which are affected as a result of the transformer leakage inductance are computed.

In case of an ideal transformer, all the parameters can be computed analytically and $\hat{V}_s^1 = \hat{V}_{st}^1$ as the transformer is ideal. For an ideal transformer, there is no phase shift between the fundamental component of link current and link voltage and hence the displacement power factor is 1. The apparent power can be computed using (8.6) and it is 603.9 kVA even though the active power is 500 kW. As the output voltage of the transformer is a multi-step voltage (close to sinusoidal), for a constant average voltage of V_{dc} , $\hat{V}_s^1 = \frac{\pi}{2} V_{dc}$, which increases the voltage rating of individual transformers.

For the second case when the transformer leakage inductance is present, \hat{V}_s^1 is determined analytically from (8.10) and compared with the simulation result. The analytical value of \hat{V}_s^1 for 50 μH is 6238.6 V and for 100 μH is 6667.9 V close to that obtained from simulation. The values for the other parameters are extracted directly from the simulation done in MATLAB environment.

The Table 8.2 clearly shows that due to the leakage inductance the link current smoothens,

resulting in a reduction of the current % THD. But the displacement power factor between \hat{I}_s^1 and \hat{V}_s^1 increases, resulting in increased reactive power losses. Hence, with the presence of transformer leakage inductance the apparent power increases to 633.04 kVA and 666.33 kVA for a transformer with 50 μH and 100 μH of total leakage inductance as compared to 603.9 kVA for an ideal transformer.

It is important to design the transformers with low leakage inductance but as the secondary side contains a diode bridge the proposed topology does not require any clamp circuit or any complex leakage commutation strategy.



8.1.5 Control of CSI

Figure 8.2: Control Diagram for a PWM-CSI

The dc link current I_{dc} in a PWM CSI supplies the active current component to the grid and also supplies the filter capacitor reactive current in case of a unity power factor operation. A closed loop control using a PI controller is implemented to maintain a constant current, I_{dc} in the steady state as shown in Fig. 8.2. Simple carrier based modulation approach is used for modulating the CSI [102]. The control scheme is employed by changing the modulation index supplied to the PWM CSI.

The *d*-axis is aligned with the phase *a* axis. Hence, the *d*-axis voltage, v_{sd} is same as the phase-to-neutral peak voltage, \hat{v}_g and *q*-axis voltage, v_{sq} is 0.

The active power in dq reference frame is:

$$P_g = 1.5 v_{sd} i_{sd} \tag{8.11}$$

The active current component can be obtained from (8.11) and the reactive current can be computed using $(2\pi f_s C_i \hat{v}_q)$ assuming that the drop across the output filter inductor is
negligible so that $\hat{v}_c = \hat{v}_g$. The reference current is computed using (8.1). The modulation index m_i is considered to be 1, its maximum value.

The error signal to the PI controller is the difference of the reference dc link current, I_{dc}^* and the actual sensed current, I_{dc} . The controller ensures that the active current component of the dc link current varies according to the required active power. The output of the controller is the modulation index given to the PWM-CSI.

8.2 Simulation results



Figure 8.3: (a) Top to Bottom: Transformer secondary voltages across each of the three transformers followed by the total secondary voltage (b) Top to Bottom: Transformer primary current, Two level current at the output of the CSI I_{csi} , Capacitor Filter current, I_c , (c) Top to Bottom: Input dc current, I_{in} , DC link current, I_{dc} , Grid voltage, V_g and Grid Current, I_g

The proposed topology was simulated using MATLAB/Simulink. The dc voltage source of $V_{in} = 300 V$ is to be integrated to a three-phase 3.3 kV/500 kW grid at 60 Hz. The high-frequency transformer to step up the low input voltage to three-phase medium voltage is to be designed at 10 kHz. The transformer leakage inductances are considered to be 0.04 pu.

The top plot in Fig. 8.3b shows the primary current of one HFT. The diode bridge connected to the transformer secondary causes a trapezoidal current through the transformer windings, hence eliminating the problem of transformer leakage commutation. The primary current waveform nature is the same in all three primary windings but the magnitude of the primary currents vary depending on the transformer turns ratio. The second plot in Fig. 8.3b is of the two level current waveform of the CSI for phase a and the third plot is of the filter capacitor current of phase a. The fundamental peak component of this capacitor current is equal to $2\pi f_s C_i \hat{v}_c$ as given in (8.1).

Fig. 8.3c shows a plot of I_{in} , I_{dc} and the grid side voltage and current. The mean value of input current, I_{in} is close to 1666.67 A, according to $I_{in} = \frac{P_g}{V_{in}}$, but the input current goes negative on an instantaneous basis as shown in the first plot of Fig. 8.3c. This will require a capacitor in the input to avoid the flow of negative current into the dc source. I_{dc} reaches a constant value of 123.71 A in steady state as the error signal to the PI control block goes to 0. In order to achieve acceptable grid current THD (< 5%) and a close to unity power factor, the grid side filter inductance is computed to be $L_{filter} = 5 \ mH$ and the filter capacitor is chosen to be $C_{filter} = 50 \ uF$.

Chapter 9

Conclusion and Future Work

9.1 Conclusion

An asymmetrical single-stage multilevel converter with HFT isolation has been proposed in this thesis for interfacing low voltage DC to high voltage AC grid. The topology combines the advantages of single-stage power conversion, galvanic isolation in a small footprint using HFT and lower voltage THD by generating multilevel voltages. The ideal modulation along with the four-step commutation algorithm for the bi-directional switches was implemented. This resulted in an output voltage profile having glitches, arising due to switch non-idealities. The cause and effect of the glitches was described in detail and a compensation technique was implemented to negate the voltage reversal. The compensation allows to improve the input current ripple profile and hence, enhances the system power quality. Another set of glitches, in the output voltage due to the interdependence of the three AC-AC converter modules were investigated and a compensation technique was proposed and demonstrated.

Another modulation technique has been investigated for the isolated asymmetrical converter using conventional level shifted carriers. Considering ideal switches, the generation of fifteen levels in the phase-to-neutral output voltage using three AC-AC converter modules is explained. However, because of the non-idealistic nature of the switch, the switching signals have to be modified. The modifications in the ideal signals to incorporate the dead-time of primary side H-bridge switches and four-step commutation of bi-directional switches is explained in detail by showing six different modes of circuit operation for one particular switching transition. The modified signals based on the high-frequency link shift the harmonic frequencies in the phase-to-neutral output voltage to the sidebands of multiples of twice the switching frequency, hence, reducing the size of the grid filter. The analysis in general can be used for any isolated single-stage converter and will also aid in the real-time implementation of the converter. The design and characterization of the multi-winding transformer to achieve low leakage inductance was explained. Simulations and experimental results on a laboratory prototype were given to demonstrate the proposed topology and validate the compensation techniques.

The topology has certain challenges in terms of: 1) higher control complexity in comparison to two-stage conversion; 2) unlike the dual active bridge, control of the proposed topology does not allow complete soft-switching; 3) the asymmetrical configuration although enables to achieve more levels with lesser device count, however, the absence of any redundant modules affects the system in case of a module failure; 4) the proposed topology requires a multi-winding transformer design which is difficult to manufacture.

The following has been done in this thesis to address those challenges: i) A normalized index value similar to [26] has been computed to determine the optimum number of levels for asymmetrical single-stage conversion in terms of voltage THD, control complexity, switch count and number of multi-winding transformer terminals. Based on the data, 1:2:4 asymmetrical configuration allows to achieve the lowest normalized index; ii) implementation of four-step commutation, results in 3/4th transitions to be soft-switched even at low currents. iii) converter operation in case of module failure is outlined by a combination of modification in control and hardware reconfiguration [95, 97]. The other single-stage HFT isolated topologies face similar challenges [45, 68], whereas the two stage topologies require reliability critical DC-link electrolytic capacitor [103] and have issues caused due to multi-stage converter. However, a quantitative analysis of the reliability of the proposed converter needs to be done similar to [104], [105] but is a part of future work.

9.2 Study of several isolated multilevel converter topologies

Table 9.1 presents a case study of several isolated multilevel converter topologies in the literature. It is to be noted that for an equivalent comparison with all the topologies, a three phase system is considered. For a power level, P, phase-to-neutral voltage, v_{pn} , input voltage, V_{in} and input current, I_{in} , a comparison of several topologies is outlined based on the voltage and the current rating of the devices and the control algorithm used.

Topology*	Modules		Device Rating		Levels	Capacitor	Switches		$\operatorname{Control}^\dagger$
	N_{pm}	N_{sm}	$V_{rs}[V]$	$I_{ps}[\mathbf{A}]$			S_p	S_s	
ISTSC [27]	9	9	$\frac{\hat{v}_{pn}}{3}$	$\frac{I_{in}}{9}$	7	Identical	36	72	PWM
ISSSC1 [80]	3	6	$\frac{\hat{v}_{pn}}{2}$	$\frac{I_{in}}{3}$	5	-	12	24	SVM
ISSSC2 [68]	8	24	$\frac{\hat{v}_{pn}}{8}$	$\frac{I_{in}}{8}$	17	-	32	192	PWM
IATSC [61]	3		V_{in}	$\frac{0.8I_{in}}{3}$	27	Unidentical	12		
	1	3	$\frac{3\hat{v}_{pn}}{13}$	$0.2I_{in}$			4	24	NLM
		3	$\frac{\hat{v}_{pn}}{13}$					24	
IASSC1 [63]	1	3	$\frac{4\hat{v}_{pn}}{7}$	I_{in}	15	-	4	24	Uni-
		3	$\frac{2\hat{v}_{pn}}{7}$					24	Polar
		3	$\frac{\hat{v}_{pn}}{7}$					24	PWM
IASSC2	3	3	$\frac{4\hat{v}_{pn}}{7}$	$\frac{I_{in}}{3}$	15	-	12	24	PWM
		3	$\frac{2\hat{v}_{pn}}{7}$					24	
		3	$\frac{\hat{v}_{pn}}{7}$					24	

Table 9.1: Case Study of Several Isolated Multilevel Converter Topologies

*ISTSC- Isolated Symmetrical Two Stage Converter; ISSSC1- Isolated Symmetrical Single Stage Converter 1 ISSSC2- Isolated Symmetrical Single Stage Converter 2; IATSC- Isolated Asymmetrical Two Stage Converter IASSC1- Isolated Asymmetrical Single Stage Converter 1; IASSC1- Isolated Asymmetrical Single Stage Converter 2 [†]PWM- Pulse Width Modulation; SVM- Space Vector Modulation; NLM- Nearest Level Modulation

ISTSC uses a two stage conversion from high-frequency AC-DC-line frequency AC and hence requires a large number of reliability critical DC-link capacitors. However, the high frequency dual active bridge configuration allows the devices to be soft-switched. ISSSC2 on the other hand, due to single stage conversion remove the DC-link capacitors. Such symmetrical multilevel converter topologies, require many modules to be connected in series to generate a multilevel voltage resulting in a large system. However, the reliability of the system can be increased by having a specific number of redundant modules.

Isolated asymmetrical converters can generate a large number of levels using less number

of devices. The switch count in Table 9.1 shows the difference in the number of devices in an asymmetrical converter in comparison to a symmetrical case. IATSC can achieve the lowest THD with the least number of devices. The diode bridges are replaced with Hbridges for bi-directional power flow and an equivalent device count. However, the absence of a transformer winding for the main converter requires a high voltage DC input and high voltage rating devices. Nearest level modulation is used to generate the 27-level output voltage. IASSC2 uses asymmetrical single stage modules to generate a multilevel output voltage using modified PWM. The topology allows to achieve fifteen levels in the output voltage with less number of switches and the single stage conversion allows to get rid of the DC-link capacitor. The modulation results in the dominant harmonics at the sidebands of multiples of twice the switching frequency resulting in a smaller grid filter. Also, by the virtue of four-step commutation in every transition 3/4th of the transitions are softswitched [106]. However, due to asymmetrical configuration the reliability of the system is a concern due to the absence of redundant modules.

9.3 Future work

- Investigation of system efficiency and loss breakdown.
- Determination of optimal switching frequency based on the loss model.
- Quantitative analysis of system reliability.
- Investigate techniques to retrieve the energy lost in the leakage inductance for gate drive power supplies etc.
- Comparison of transformer design using planar, non-planar geometry and using ferrite or amorphous/nanocrystalline material in terms of leakage inductance.
- Detailed investigation of a proposed topology consisting of full-bridge modules in parallel on the low voltage input side and Modular Multi-level Converter(MMCs) on the high voltage output side with a series of asymmetrical two-winding transformers.

References

- J.M. Carrasco, L.G. Franquelo, J.T. Bialasiewicz, E. Galvan, R.C.P. Guisado, Ma.A.M. Prats, J.I. Leon, and N. Moreno-Alfonso. Power-electronic systems for the grid integration of renewable energy sources: A survey. *Industrial Electronics, IEEE Transactions on*, 53(4):1002–1016, June 2006.
- [2] K. Euerle, K. Iyer, E. Severson, R. Baranwal, S. Tewari, and N. Mohan. A compact active filter to eliminate common-mode voltage in a sic-based motor drive. In 2016 IEEE Energy Conversion Congress and Exposition (ECCE), pages 1–8, Sept 2016.
- [3] S. Song, F. Kang, and S. J. Park. Cascaded multilevel inverter employing threephase transformers and single dc input. *IEEE Transactions on Industrial Electronics*, 56(6):2005–2014, June 2009.
- [4] A. K. Adnanes. Maritime electrical installations and diesel electric propulsion, 2003.
- [5] N. H. Doerry and J. C. Davis. Integrated power system for marine applications. In Naval Engineers Journal, pages 77–90, 1994.
- [6] J.S. Thongam, M. Tarbouchi, A.F. Okou, D. Bouchard, and R. Beguenane. All-electric ships; a review of the present state of the art. In *Ecological Vehicles and Renewable Energies (EVER), 2013 8th International Conference and Exhibition on*, pages 1–8, March 2013.
- [7] J.S. Chalfant and C. Chryssostomidis. Analysis of various all-electric-ship electrical distribution system topologies. In *Electric Ship Technologies Symposium (ESTS)*, 2011 IEEE, pages 72–77, April 2011.
- [8] J.M. Apsley, A. Gonzalez-Villasenor, M. Barnes, A.C. Smith, S. Williamson, J.D. Schuddebeurs, P.J. Norman, C.D. Booth, G.M. Burt, and J.R. McDonald. Propulsion drive models for full electric marine propulsion systems. *Industry Applications, IEEE Transactions on*, 45(2):676–684, March 2009.

- [9] J.F. Hansen and F. Wendt. History and state of the art in commercial electric ship propulsion, integrated power systems, and future trends. *Proceedings of the IEEE*, 103(12):2229–2242, Dec 2015.
- [10] J.M. Prousalidis, G.J. Tsekouras, and F. Kanellos. New challenges emerged from the development of more efficient electric energy generation units. In *Electric Ship Technologies Symposium (ESTS), 2011 IEEE*, pages 374–381, April 2011.
- [11] R.M. Cuzner, V. Singh, M. Rashidi, and A. Nasiri. Converter topological and solid state protective device trade-offs for future shipboard mvdc systems. In *Electric Ship Technologies Symposium (ESTS), 2015 IEEE*, pages 34–39, June 2015.
- [12] S. Kulkarni and S. Santoso. Estimating transient response of simple ac and dc shipboard power systems to pulse load operations. In *Electric Ship Technologies Sympo*sium, 2009. ESTS 2009. IEEE, pages 73–78, April 2009.
- [13] R. Nilsen and I. Sorfonn. Hybrid power generation systems. In Power Electronics and Applications, 2009. EPE '09. 13th European Conference on, pages 1–9, Sept 2009.
- [14] So-Yeon Kim, Byung-Geuk Cho, and Seung-Ki Sul. Feasibility study of integrated power system with battery energy storage system for naval ships. In Vehicle Power and Propulsion Conference (VPPC), 2012 IEEE, pages 532–537, Oct 2012.
- [15] C. Heising, V. Staudt, and A. Steimel. Optimized energy-efficient drive system for ship propulsion. In *Electric Ship Technologies Symposium (ESTS)*, 2011 IEEE, pages 292–295, April 2011.
- [16] B. Zahedi and L.E. Norum. Modeling and simulation of all-electric ships with low-voltage dc hybrid power systems. *Power Electronics, IEEE Transactions on*, 28(10):4525–4537, Oct 2013.
- [17] S. De Breucker, E. Peeters, and J. Driesen. Possible applications of plug-in hybrid electric ships. In *Electric Ship Technologies Symposium*, 2009. ESTS 2009. IEEE, pages 310–317, April 2009.
- [18] So-Yeon Kim, Sehwa Choe, Sanggi Ko, and Seung-Ki Sul. A naval integrated power system with a battery energy storage system: Fuel efficiency, reliability, and quality of power. *Electrification Magazine*, *IEEE*, 3(2):22–33, June 2015.
- [19] A. Bidram, A. Davoudi, and F.L. Lewis. Distributed control for ac shipboard power systems. In *Electric Ship Technologies Symposium (ESTS)*, 2013 IEEE, pages 282– 286, April 2013.

- [20] United States Coast Guard, http://www.uscg.mil/pacarea/cgchealy/ship.asp. Enclosure(203).
- [21] R. Bucknall and K.M. Ciaramella. On the conceptual design and performance of a matrix converter for marine electric propulsion. *Power Electronics, IEEE Transactions* on, 25(6):1497–1508, June 2010.
- [22] S. Kuznetsov. Advanced ac to ac power conversion techniques for naval ship propulsion drives. In *Electric Ship Technologies Symposium (ESTS)*, 2011 IEEE, pages 420–423, April 2011.
- [23] Xu-Jing Tang, Tian Wang, Chen Zhi, and Ye-Mao Huang. The design of power management system for solar ship. In *Transportation Information and Safety (ICTIS)*, 2015 International Conference on, pages 548–553, June 2015.
- [24] T. Dragicevic, Xiaonan Lu, J. C. Vasquez, and J. M. Guerrero. Dc microgrids 2014;part ii: A review of power architectures, applications, and standardization issues. *IEEE Transactions on Power Electronics*, 31(5):3528–3549, May 2016.
- [25] Xu She, A.Q. Huang, and R. Burgos. Review of solid-state transformer technologies and their application in power distribution systems. *Emerging and Selected Topics in Power Electronics, IEEE Journal of*, 1(3):186–198, Sept 2013.
- [26] M.R. Islam, Youguang Guo, and Jianguo Zhu. A high-frequency link multilevel cascaded medium-voltage converter for direct grid integration of renewable energy systems. *Power Electronics, IEEE Transactions on*, 29(8):4167–4182, Aug 2014.
- [27] Biao Zhao, Qiang Song, and Wenhua Liu. A practical solution of high-frequencylink bidirectional solid-state transformer based on advanced components in hybrid microgrid. *Industrial Electronics, IEEE Transactions on*, 62(7):4587–4597, July 2015.
- [28] S. Madhusoodhanan, K. Hatua, S. Bhattacharya, S. Leslie, Sei-Hyung Ryu, M. Das, A. Agarwal, and D. Grider. Comparison study of 12kv n-type sic igbt with 10kv sic mosfet and 6.5kv si igbt based on 3l-npc vsc applications. In *Energy Conversion Congress and Exposition (ECCE)*, 2012 IEEE, pages 310–317, Sept 2012.
- [29] R. Otero-De-Leon and N. Mohan. A push-pull dc-ac high frequency power electronics transformer for photovoltaic applications. In *IECON 2014 - 40th Annual Conference* of the *IEEE Industrial Electronics Society*, pages 1131–1136, Oct 2014.
- [30] R. Otero-De-Leon and N. Mohan. Full-bridge modular multilevel converter with high frequency link for photovoltaic applications. In 2015 IEEE 24th International Symposium on Industrial Electronics (ISIE), pages 294–299, June 2015.

- [31] K. V. Iyer and N. Mohan. Asymmetrical converters with medium frequency transformer isolation to interface low voltage dc to a medium voltage three phase grid. In 2014 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), pages 1–6, Dec 2014.
- [32] H.S. Krishnamoorthy, S. Essakiappan, P.N. Enjeti, R.S. Balog, and S. Ahmed. A new multilevel converter for megawatt scale solar photovoltaic utility integration. In *Applied Power Electronics Conference and Exposition (APEC), 2012 Twenty-Seventh Annual IEEE*, pages 1431–1438, Feb 2012.
- [33] M. Amirabadi, A. Balakrishnan, H.A. Toliyat, and W.C. Alexander. High-frequency ac-link pv inverter. *Industrial Electronics, IEEE Transactions on*, 61(1):281–291, Jan 2014.
- [34] R. Baranwal, G. F. Castelino, and N. Mohan. Power electronic transformer with reduced switch count and dual matrix converters to eliminate load common-mode voltage and input snubber requirement. In 2014 International Symposium on Power Electronics, Electrical Drives, Automation and Motion, pages 769–774, June 2014.
- [35] K. Basu, A. Shahani, A. K. Sahoo, and N. Mohan. A single-stage solid-state transformer for pwm ac drive with source-based commutation of leakage energy. *IEEE Transactions on Power Electronics*, 30(3):1734–1746, March 2015.
- [36] A. K. Sahoo and N. Mohan. High frequency link multi-winding power electronic transformer using modular multilevel converter for renewable energy integration. In *IECON 2014 - 40th Annual Conference of the IEEE Industrial Electronics Society*, pages 4642–4648, Oct 2014.
- [37] S. Gandikota and N. Mohan. A new leakage energy commutation technique for single stage high frequency link inverters. In 2014 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), pages 1–6, Dec 2014.
- [38] S. Gandikota and N. Mohan. High frequency ac-link transformer isolated three phase inverter with natural commutation of leakage energy. In 7th IET International Conference on Power Electronics, Machines and Drives (PEMD 2014), pages 1–5, April 2014.
- [39] B. Zhao, Q. Song, W. Liu, and Y. Sun. Overview of dual-active-bridge isolated bidirectional dc-dc converter for high-frequency-link power-conversion system. *IEEE Transactions on Power Electronics*, 29(8):4091–4106, Aug 2014.

- [40] C. Zhao, S. D. Round, and J. W. Kolar. An isolated three-port bidirectional dcdc converter with decoupled power flow management. *IEEE Transactions on Power Electronics*, 23(5):2443–2453, Sept 2008.
- [41] A. K. Sahoo and N. Mohan. A power electronic transformer with sinusoidal voltages and currents using modular multilevel converter. In 2014 International Power Electronics Conference (IPEC-Hiroshima 2014 - ECCE ASIA), pages 3750–3757, May 2014.
- [42] B. Zhao, Q. Song, J. Li, Y. Wang, and W. Liu. High-frequency-link modulation methodology of dc-dc transformer based on modular multilevel converter for hvdc application: Comprehensive analysis and experimental verification. *IEEE Transactions* on Power Electronics, PP(99):1–1, 2016.
- [43] C. Zhao, S. D. Round, and J. W. Kolar. An isolated three-port bidirectional dcdc converter with decoupled power flow management. *IEEE Transactions on Power Electronics*, 23(5):2443–2453, Sept 2008.
- [44] R. Baranwal, G. Castelino, K. V. Iyer, K. Basu, and N. Mohan. A dual active bridge based single phase ac to dc power electronic transformer with advanced features. *IEEE Transactions on Power Electronics*, PP(99):1–1, 2017.
- [45] H. Qin and J. W. Kimball. Solid-state transformer architecture using ac-ac dualactive-bridge converter. *IEEE Transactions on Industrial Electronics*, 60(9):3720– 3730, Sept 2013.
- [46] D. Chen and Y. Chen. Step-up ac voltage regulators with high-frequency link. IEEE Transactions on Power Electronics, 28(1):390–397, Jan 2013.
- [47] M. Kim, M. Rosekeit, S. K. Sul, and R. W. A. A. De Doncker. A dual-phase-shift control strategy for dual-active-bridge dc-dc converter in wide voltage range. In *Power Electronics and ECCE Asia (ICPE ECCE), 2011 IEEE 8th International Conference* on, pages 364–371, May 2011.
- [48] N. Hou, W. Song, and M. Wu. Minimum-current-stress scheme of dual active bridge dc-dc converter with unified phase-shift control. *IEEE Transactions on Power Electronics*, 31(12):8552–8561, Dec 2016.
- [49] F. Krismer, J. Biela, and J. W. Kolar. A comparative evaluation of isolated bidirectional dc/dc converters with wide input and output voltage range. In *Fourtieth IAS Annual Meeting. Conference Record of the 2005 Industry Applications Conference, 2005.*, volume 1, pages 599–606 Vol. 1, Oct 2005.

- [50] J. Wang, G. Wang, S. Bhattacharya, and A. Q. Huang. Comparison of 10-kv sic power devices in solid-state transformer. In *Energy Conversion Congress and Exposition* (ECCE), 2010 IEEE, pages 3284–3289, Sept 2010.
- [51] A. K. Sahoo, A. Shahani, K. Basu, and N. Mohan. Lcl filter design for grid-connected inverters by analytical estimation of pwm ripple voltage. In 2014 IEEE Applied Power Electronics Conference and Exposition - APEC 2014, pages 1281–1286, March 2014.
- [52] Weimin Wu, Yunjie Sun, Min Huang, Xiongfei Wang, Huai Wang, F. Blaabjerg, M. Liserre, and H.S.-H. Chung. A robust passive damping method for llcl-filter-based grid-tied inverters to minimize the effect of grid harmonic voltages. *Power Electronics*, *IEEE Transactions on*, 29(7):3279–3289, July 2014.
- [53] A Rufer, M Veenstra, and K Gopakumar. Asymmetric multilevel converter for high resolution voltage phasor generation. In *EPE99*, pages P1–P10, 1999.
- [54] Yu Liu and Fang-Lin Luo. Trinary hybrid 81-level multilevel inverter for motor drive with zero common-mode voltage. *Industrial Electronics, IEEE Transactions* on, 55(3):1014–1021, March 2008.
- [55] J. Dixon, A.A. Breton, F.E. Rios, J. Rodriguez, J. Pontt, and M.A. Perez. High-power machine drive, using nonredundant 27-level inverters and active front end rectifiers. *Power Electronics, IEEE Transactions on*, 22(6):2527–2533, Nov 2007.
- [56] E. Samadaei, S. Gholamian, A. Sheikholeslami, and J. Adabi. An envelope type (e-type) module: Asymmetric multilevel inverters with reduced components. *IEEE Transactions on Industrial Electronics*, PP(99):1–1, 2016.
- [57] S. K. Chattopadhyay and C. Chakraborty. Performance of three-phase asymmetric cascaded bridge (16:4:1) multilevel inverter. *IEEE Transactions on Industrial Electronics*, 62(10):5983–5992, Oct 2015.
- [58] K. V. Iyer, A. K. Sahoo, and N. Mohan. Asymmetrical modular mutilevel converter (ammc) based hvdc system. In 2015 IEEE Energy Conversion Congress and Exposition (ECCE), pages 4693–4699, Sept 2015.
- [59] Feel-Soon Kang, Sung-Jun Park, Man-Hyung Lee, and Cheul-U Kim. An efficient multilevel-synthesis approach and its application to a 27-level inverter. *Industrial Electronics, IEEE Transactions on*, 52(6):1600–1606, Dec 2005.
- [60] P. Barriuso, J. Dixon, P. Flores, and L. Moran. Fault-tolerant reconfiguration system for asymmetric multilevel converters using bidirectional power switches. *IEEE Transactions on Industrial Electronics*, 56(4):1300–1306, April 2009.

- [61] J. Pereda and J. Dixon. High-frequency link: A solution for using only one dc source in asymmetric cascaded multilevel inverters. *Industrial Electronics, IEEE Transactions* on, 58(9):3884–3892, Sept 2011.
- [62] J. Dixon, J. Pereda, C. Castillo, and S. Bosch. Asymmetrical multilevel inverter for traction drives using only one dc supply. *Vehicular Technology, IEEE Transactions* on, 59(8):3736–3743, Oct 2010.
- [63] K. V. Iyer and N. Mohan. Multi-level converter to interface low voltage dc to 3-phase high voltage grid with medium frequency transformer isolation. In *IECON 2014 -*40th Annual Conference of the IEEE Industrial Electronics Society, pages 4683–4689, Oct 2014.
- [64] K. V. Iyer, R. Baranwal, and N. Mohan. A high-frequency ac-link single-stage asymmetrical multilevel converter for grid integration of renewable energy systems. *IEEE Transactions on Power Electronics*, PP(99):1–1, 2016.
- [65] K. V. Iyer and N. Mohan. Modulation and commutation of a single stage isolated asymmetrical multilevel converter for the integration of renewables and battery energy storage system in ships. *IEEE Transactions on Transportation Electrification*, 2(4):580–596, Dec 2016.
- [66] S. Kouro, M. Malinowski, K. Gopakumar, J. Pou, L.G. Franquelo, Bin Wu, J. Rodriguez, M.A. Perez, and J.I. Leon. Recent advances and industrial applications of multilevel converters. *Industrial Electronics, IEEE Transactions on*, 57(8):2553–2580, Aug 2010.
- [67] M. Perez, J. Rodriguez, J. Pontt, and S. Kouro. Power distribution in hybrid multicell converter with nearest level modulation. In *Industrial Electronics*, 2007. ISIE 2007. IEEE International Symposium on, pages 736–741, June 2007.
- [68] S. Essakiappan, H.S. Krishnamoorthy, P. Enjeti, R.S. Balog, and S. Ahmed. Multilevel medium-frequency link inverter for utility scale photovoltaic integration. *Power Electronics, IEEE Transactions on*, 30(7):3674–3684, July 2015.
- [69] Richard Lund, Madhav D Manjrekar, Peter Steimer, and Thomas A Lipo. Control strategies for a hybrid seven-level inverter. In *Proceedings of the European Power Electronic Conference*, 1999.
- [70] Z. Chen, Z. Wang, Y. Liu, and Y. Luo. A unipolar pwm strategy for hybrid cascaded multilevel converters. In *IECON 2014 - 40th Annual Conference of the IEEE Industrial Electronics Society*, pages 1154–1160, Oct 2014.

- [71] Yi Wang, Heming Li, Xinchun Shi, and Boqiang Xu. A novel carrier-based pwm strategy for hybrid multilevel inverters. In *Power Electronics Specialists Conference*, 2004. PESC 04. 2004 IEEE 35th Annual, volume 6, pages 4233–4237 Vol.6, June 2004.
- [72] T. Schulte and G. Schrder. Power loss comparison of different matrix converter commutation strategies. In *Power Electronics and Motion Control Conference* (*EPE/PEMC*), 2012 15th International, pages DS2c.9–1–DS2c.9–6, Sept 2012.
- [73] N. Burany. Safe control of four-quadrant switches. In Industry Applications Society Annual Meeting, 1989., Conference Record of the 1989 IEEE, pages 1190–1194 vol.1, Oct 1989.
- [74] M. Ziegler and W. Hofmann. Semi natural two steps commutation strategy for matrix converters. In *Power Electronics Specialists Conference*, 1998. PESC 98 Record. 29th Annual IEEE, volume 1, pages 727–731 vol.1, May 1998.
- [75] M. Ziegler and Hofmann. New one-step commutation strategies in matrix converters. In Power Electronics and Drive Systems, 2001. Proceedings., 2001 4th IEEE International Conference on, volume 2, pages 560–564 vol.2, Oct 2001.
- [76] L. C. Herrero, S. de Pablo, F. Martin, J. M. Ruiz, J. M. Gonzalez, and A. B. Rey. Comparative analysis of the techniques of current commutation in matrix converters. In 2007 IEEE International Symposium on Industrial Electronics, pages 521–526, June 2007.
- [77] Huu-Nhan Nguyen and Hong-Hee Lee. A modulation scheme for matrix converters with perfect zero common-mode voltage. *Power Electronics, IEEE Transactions on*, PP(99):1–1, 2015.
- [78] R. Baranwal, K. Basu, A. K. Sahoo, and N. Mohan. A modified four-step commutation to suppress common-mode voltage during commutations in open-end winding matrix converter drives. In 2015 IEEE Energy Conversion Congress and Exposition (ECCE), pages 4455–4462, Sept 2015.
- [79] S. Harb, H. Zhang, and R. S. Balog. Ac-link, single-phase, photovoltaic module integrated inverter. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 177–182, March 2013.
- [80] J. Beristain, J. Bordonau, Gilabert A, and S. Alepuz. A new ac/ac multilevel converter for a single-phase inverter with hf isolation. In *Power Electronics Specialists*

Conference, 2004. PESC 04. 2004 IEEE 35th Annual, volume 3, pages 1998–2004 Vol.3, June 2004.

- [81] K. V. Iyer, K. Basu, W. P. Robbins, and N. Mohan. Determination of the optimal thickness for a multi-layer transformer winding. In 2013 IEEE Energy Conversion Congress and Exposition, pages 3738–3741, Sept 2013.
- [82] G. Ortiz, J. Biela, and J.W. Kolar. Optimized design of medium frequency transformers with high isolation requirements. In *IECON 2010 - 36th Annual Conference* on *IEEE Industrial Electronics Society*, pages 631–638, 2010.
- [83] Y. A. Wang, D. M. Xiao, and Y. L. Liu. Design of a planar power transformer for high voltage, high frequency use. In *IEEE PES T D 2010*, pages 1–6, April 2010.
- [84] Z. Ouyang and M. A. E. Andersen. Overview of planar magnetic technology 2014;fundamental properties. *IEEE Transactions on Power Electronics*, 29(9):4888–4900, Sept 2014.
- [85] Md. Rabiul Islam, Youguang Guo, Zhi Wei Lin, and Jianguo Zhu. An amorphous alloy core medium frequency magnetic-link for medium voltage photovoltaic inverters. *Journal of Applied Physics*, 115(17), 2014.
- [86] M. R. Islam, G. Lei, Y. Guo, and J. Zhu. Optimal design of high-frequency magnetic links for power converters used in grid-connected renewable energy systems. *IEEE Transactions on Magnetics*, 50(11):1–4, Nov 2014.
- [87] K. Mainali, A. Tripathi, S. Madhusoodhanan, A. Kadavelugu, D. Patel, S. Hazra, K. Hatua, and S. Bhattacharya. A transformerless intelligent power substation: A three-phase sst enabled by a 15-kv sic igbt. *IEEE Power Electronics Magazine*, 2(3):31–43, Sept 2015.
- [88] A. Maitra, A. Sundaram, M. Gandhi, S. Bird, and S. Doss. Intelligent universal transformer design and applications. In *Electricity Distribution - Part 1, 2009. CIRED* 2009. 20th International Conference and Exhibition on, pages 1–7, June 2009.
- [89] Ned Mohan, Tore M Undeland, and William P Robbins. Power electronics: converters, applications, and design. 2003.
- [90] K. V. Iyer, W. P. Robbins, and N. Mohan. Design and comparison of high frequency transformers using foil and round windings. In 2014 International Power Electronics Conference (IPEC-Hiroshima 2014 - ECCE ASIA), pages 3037–3043, May 2014.

- [91] K. V. Iyer, W. P. Robbins, and N. Mohan. Winding design of a high power medium frequency transformer. In 2014 International Symposium on Power Electronics, Electrical Drives, Automation and Motion, pages 665–669, June 2014.
- [92] K. V. Iyer, W. P. Robbins, K. Basu, and N. Mohan. Transformer winding losses with round conductors for duty-cycle regulated square waves. In 2014 International Power Electronics Conference (IPEC-Hiroshima 2014 - ECCE ASIA), pages 3061–3066, May 2014.
- [93] Kartik Iyer. Transformer winding losses with round conductors and foil windings for duty-cycle regulated square waveform followed by winding design and comparison for sinusoidal excitation. Master's thesis, UNIVERSITY OF MINNESOTA, 2013.
- [94] R.D. Middlebrook and S. Cuk. Advances in switched-mode power conversion, volumes 1-2. TESLAco, 1983.
- [95] W. Zhang, D. Xu, P. N. Enjeti, H. Li, J. T. Hawke, and H. S. Krishnamoorthy. Survey on fault-tolerant techniques for power electronic converters. *IEEE Transactions on Power Electronics*, 29(12):6319–6331, Dec 2014.
- [96] P. Lezana and G. Ortiz. Extended operation of cascade multicell converters under fault condition. *IEEE Transactions on Industrial Electronics*, 56(7):2697–2703, July 2009.
- [97] P. Barriuso, J. Dixon, P. Flores, and L. Moran. Fault-tolerant reconfiguration system for asymmetric multilevel converters using bidirectional power switches. *IEEE Transactions on Industrial Electronics*, 56(4):1300–1306, April 2009.
- [98] Magnetics Inc, http://www.mag-inc.com/products/powder-cores/permeabilityversus-dc-bias-curves-for-powder-cores. *Permeability vs DC Bias curve*.
- [99] Changsung Corporation, http://www.mhwintl.com/assets/CSC/CSC Catalog.pdf. CSC Catalog.
- [100] EPCOS, http://en.tdk.eu/blob/519704/download/2/ferrites-and-accessories-databook-130501.pdf. EPCOS Databook(2013).
- [101] Xiaotian Tan, Jingya Dai, and Bin Wu. A novel converter configuration for wind applications using pwm csi with diode rectifier and buck converter. In *Electric Machines Drives Conference (IEMDC), 2011 IEEE International*, pages 359–364, May 2011.

- [102] K. Basu, A.K. Sahoo, V. Chandrasekaran, and N. Mohan. Grid-side ac line filter design of a current source rectifier with analytical estimation of input current ripple. *Power Electronics, IEEE Transactions on*, 29(12):6394–6405, Dec 2014.
- [103] H. Wang, M. Liserre, and F. Blaabjerg. Toward reliable power electronics: Challenges, design tools, and opportunities. *IEEE Industrial Electronics Magazine*, 7(2):17–26, June 2013.
- [104] M. Aten, G. Towers, C. Whitley, P. Wheeler, J. Clare, and K. Bradley. Reliability comparison of matrix and other converter topologies. *IEEE Transactions on Aerospace* and Electronic Systems, 42(3):867–875, July 2006.
- [105] A. H. Ranjbar, B. Abdi, G. B. Gharehpetian, and B. Fahimi. Reliability assessment of single-stage/two-stage pfc converters. In 2009 Compatibility and Power Electronics, pages 253–257, May 2009.
- [106] R. Moghe, R. P. Kandula, A. Iyer, and D. Divan. Losses in medium-voltage megawattrated direct ac/ac power electronics converters. *IEEE Transactions on Power Electronics*, 30(7):3553–3562, July 2015.